# OECD DEVELOPMENT CENTRE

Working Paper No. 286

# DOES TRADE STIMULATE INNOVATION? EVIDENCE FROM FIRM-PRODUCT DATA

by

Ana Margarida Fernandes and Caroline Paunov

Research area: InnovaLatino







# DEVELOPMENT CENTRE WORKING PAPERS

This series of working papers is intended to disseminate the Development Centre's research findings rapidly among specialists in the field concerned. These papers are generally available in the original English or French, with a summary in the other language.

Comments on this paper would be welcome and should be sent to the OECD Development Centre, 2, rue André Pascal, 75775 PARIS CEDEX 16, France; or to <a href="mailto:dev.contact@oecd.org">dev.contact@oecd.org</a>. Documents may be downloaded from: <a href="http://www.oecd.org/dev/wp">http://www.oecd.org/dev/wp</a> or obtained via e-mail (dev.contact@oecd.org).



THE OPINIONS EXPRESSED AND ARGUMENTS EMPLOYED IN THIS DOCUMENT ARE THE SOLE RESPONSIBILITY OF THE AUTHORS AND DO NOT NECESSARILY REFLECT THOSE OF THE OECD OR OF THE GOVERNMENTS OF ITS MEMBER COUNTRIES

©OECD (2010)

Applications for permission to reproduce or translate all or part of this document should be sent to <a href="rights@oecd.org">rights@oecd.org</a>.

# CENTRE DE DÉVELOPPEMENT DOCUMENTS DE TRAVAIL

Cette série de documents de travail a pour but de diffuser rapidement auprès des spécialistes dans les domaines concernés les résultats des travaux de recherche du Centre de développement. Ces documents ne sont disponibles que dans leur langue originale, anglais ou français ; un résumé du document est rédigé dans l'autre langue.

Tout commentaire relatif à ce document peut être adressé au Centre de développement de l'OCDE, 2, rue André Pascal, 75775 PARIS CEDEX 16, France; ou à <u>dev.contact@oecd.org</u>. Les documents peuvent être téléchargés à partir de: <a href="http://www.oecd.org/dev/wp">http://www.oecd.org/dev/wp</a> ou obtenus via le mél (dev.contact@oecd.org).



LES IDÉES EXPRIMÉES ET LES ARGUMENTS AVANCÉS DANS CE DOCUMENT SONT CEUX DES AUTEURS ET NE REFLÈTENT PAS NÉCESSAIREMENT CEUX DE L'OCDE OU DES GOUVERNEMENTS DE SES PAYS MEMBRES

©OCDE (2010)

Les demandes d'autorisation de reproduction ou de traduction de tout ou partie de ce document devront être envoyées à <u>rights@oecd.org</u>.

# **TABLE OF CONTENTS**

ACKNOWLEDGEMENTS	4
PREFACE	5
RÉSUMÉ	6
ABSTRACT	7
I. INTRODUCTION	8
II. DATA	13
III. EMPIRICAL FRAMEWORK	16
IV. RESULTS	24
V. ALTERNATIVE INTERPRETATIONS	37
VI. CONCLUSIONS AND POLICY RELEVANCE	42
ANNEXES	43
REFERENCES	48
OTHER TITLES IN THE SERIES/ AUTRES TITRES DANS LA SÉRIE	50

#### **ACKNOWLEDGEMENTS**

The authors would like to thank Richard Disney, Jonathan Haskel, Beata Javorcik, Jacques Mairesse, and seminar participants at the Edinburgh Workshop on Firm-Level Data Analysis in Transition and Developing Economies, the Third Conference on Micro Evidence on Innovation in Developing Economies, the Chicago Fed, the 4th ISGEP Conference at the University of Valencia, the University of Texas at Arlington, and the World Bank for valuable comments. The authors thank Eric Verhoogen for sharing his data on the advertising intensity of US industries and Francis Ng for help with COMTRADE data. The findings expressed in this paper are those of the authors and do not necessarily represent the views of the World Bank or the OECD, its Development Centre or their member countries.

Mailing addresses: OECD Development Centre, 2, rue André Pascal, 75775 Paris Cedex 16, France. Tel: (33-1) 4524-8215, Fax: (33-1) 4430-6150 and The World Bank, Development Research Group, 1818 H Street NW, Washington DC, 20433, E-mail: <a href="mailto:afernandes@worldbank.org">afernandes@worldbank.org</a> and <a href="mailto:caroline.paunov@oecd.org">caroline.paunov@oecd.org</a>.

#### **PREFACE**

What kinds of innovation policies make sense in emerging economies to ensure innovation contributes to economic development? Is it Costa Rican scientist Franklin Chang Díaz's plasma rocket, or the incremental efforts of thousands of small and medium enterprises to innovate based on the technological and organisational advances that have been developed elsewhere? These are some of the important questions posed by the InnovaLatino initiative. This paper, by the Development Centre's Caroline Paunov and Ana Margarida Fernandes of the World Bank, is part of our effort to answer these big questions, and serves as a background paper to the first project report. The authors answer some of these questions in the context of the newest OECD member country: Chile.

Over the past two decades, globalisation and the increased exposure to import competition has created a new economic environment for manufacturing producers in emerging economies. The authors remind us that incremental innovations – not only the rocket-science type – are particularly relevant to economic progress in a developing-country setting. This paper shows that incremental innovation – upgrading and differentiating one's products in response to increased competition from imported goods – is one way for producers to position themselves in domestic and international markets.

The paper's results show that tougher import competition does have a positive, significant, and robust impact on incremental innovation, as reflected in product-quality upgrading by Chilean firms. To the extent that these findings can be generalised to other emerging economies, they suggest that increased exposure to imports can be beneficial for innovation outcomes. Moreover, the paper finds that the mechanism driving this outcome is that firms react to the import pressure by innovating so as to differentiate their products as a way to escape competition. The findings, therefore, point to the importance of competition policy. In addition, the results indicate that easier access to imported inputs also has beneficial effects on innovation, which highlights the importance of learning from trade in stimulating innovation. However, the evidence also suggests that such benefits only arise if the right conditions hold: notably, firms must have skilled workers, and not all industries offer the same opportunities for this kind of innovative response. For economies to capture the dynamic benefits of trade the framework conditions matter.

Javier Santiso
Director and Chief Development Economist
OECD Development Centre
January 2010

# **RÉSUMÉ**

Cet article examine si le fait d'être en compétition avec des produits d'importation affecte la décision des entreprises de s'engager dans l'innovation incrémentale, qui se reflète dans une amélioration de la qualité des produits. Les résultats économétriques sont fondés sur de riches données chiliennes d'entreprises et de leurs produits. La qualité des produits est mesurée par leur valeur unitaire (leur prix) et le coût de transport des importations est utilisé comme une mesure exogène du degré de compétition des importations. Les estimations montrent un effet négatif du coût de transport sur la qualité des produits. L'article démontre que les incréments dans la valeur unitaire reflètent les incréments dans la qualité des produits et que la compétition explique nos résultats. Un accès plus aisé aux produits intermédiaires participe également à une amélioration de la qualité des produits.

**Mots clés :** Compétition des importations, coût de transport, innovation incrémentale, qualité des produits, valeur unitaire de la production, données d'entreprises, entreprises multi-produits, Chili.

Classification JEL: 031, F14, L6.

#### **ABSTRACT**

This paper investigates whether increased import competition leads firms to engage in incremental innovation reflected in product quality upgrading. The econometric analysis relies on a rich dataset of Chilean manufacturing firms and their products. Product quality is measured with unit values (prices) and imports' transport costs are used as an exogenous proxy for import competition. The estimates show a negative effect of transport costs on product quality. The paper provides explicit evidence that estimated increases in unit values capture product quality upgrading, and that competition effects of imports explain our results. Easier access to intermediate inputs also fosters quality upgrading.

**Keywords:** import competition, transport costs, incremental innovation, product quality, output unit values, firm-level data, multi-product firms, Chile

JEL Classification: 031, F14, L6.

#### I. INTRODUCTION

What are the dynamic effects of globalisation for manufacturing producers in emerging economies? The evidence on trade and growth at the micro level has focused nearly exclusively on the effects of trade liberalisation on firm productivity levels and growth (Tybout, 2000). There are, however, other important dimensions along which firms adjust to globalisation. Innovation is a major factor since it plays a crucial role for growth and welfare (Grossman and Helpman, 1991; Aghion and Howitt, 1998). Within the range of possible types of innovation, minor or incremental innovations have been shown to be important drivers of growth, particularly in emerging economies (Puga and Trefler, 2009). This paper addresses the dynamic effects of globalisation by exploring whether increased exposure to imports stimulates incremental innovation reflected in product quality upgrading for firms in an emerging economy. In order to render the evidence useful for policy design we also need to understand what mechanism explains quality upgrading. The hypothesis that we explore is that competition drives the effects: firms upgrade the quality of their products so as to differentiate them from those of their competitors.

While the idea of linking import competition to product quality upgrading is appealing, its empirical implementation faces two challenges. The first challenge concerns the measurement of product quality. To address it, we exploit a new dataset including rich information from census data on all the products manufactured by all Chilean firms during the 1997-2003 period and use unit values (prices) of products to measure their unobserved quality or sophistication. Increases in unit values correspond well to the definition of incremental innovation in the OECD Oslo manual (1997) which covers "existing product[s] whose performance has been significantly enhanced or upgraded". The empirical trade literature takes for granted the idea that increases in unit values represent improvements in unobserved quality (Fontagné and Freudenberg, 1997). For example, Kugler and Verhoogen (2008), Iacovone and Javorcik (2008) and Lelarge and Nefussi (2008) use data on unit values of domestic and exported products to proxy for product or export quality at the plant level, while Brooks (2006), Kiyota (2008) and Schott (2008) use data on exports' unit values to proxy for export quality at the industry or country level. The role of product pricing as a signal for quality has also been examined by an extensive industrial organisation literature (e.g. Thomas et al., 1998; Fluet and Garella, 2002). The market for 'lemons' of Akerlof (1970) illustrates this clearly: in the presence of imperfect information, firms with high quality products need to introduce signals - higher prices - to convey to consumers the high quality of their products. Khandelwal (2009) criticizes the assumption that unit values equal

<sup>1.</sup> Pietrobelli and Rabellotti (2006) also equate product quality upgrading with "innovation to increase value-added" incremental innovation.

quality as being too strong as prices reflect also variations in costs but shows that the mapping of unit values to quality is actually appropriate in the industries with more scope for quality differentiation. In any case, in spite of the many arguments supporting the idea that unit values proxy for quality, we investigate the assumption empirically.

The second challenge concerns the difficulty in identifying causal effects of import competition on quality upgrading as upgrading can itself affect whether and how much foreign competitors choose to export to the domestic market. This challenge is particularly relevant in the case of our sample period for Chile during which a gradual and continuing process of trade integration – rather than a radical trade liberalisation episode – was under way. To address it, we rely on an effective trade barrier measure - imports' transport costs - which proxies for differences in import competition across industries that are exogenous to quality upgrading. While we are not the first to use imports' transport costs, we demonstrate why that measure is indeed a useful proxy for import competition in general and in the specific case of Chile. Our exercise contributes to the literature by providing evidence on the impact of trade openness on innovation beyond the initial effects of radical liberalisation. Our econometric approach exploits the variation in imports' transport costs across 4-digit industries and over time and consists of regressions of product unit values on a lagged measure of transport costs, firm-product and industry-year fixed effects, and firm and industry control variables. Importantly, our specifications identify the impact of transport costs by comparing unit values of a given product within a firm as transport costs change.

Our main finding is that stronger import competition leads to significant quality upgrading by Chilean firms. This result is obtained in regressions that use transport costs measures directly and is also confirmed in regressions that use import penetration ratios instrumented by transport costs measures. We prove that our estimated increases in unit values due to tougher import competition result from product quality improvements by showing that there is no impact of import competition from less advanced economies than Chile whose product quality is likely insufficient to challenge Chilean producers. Moreover, we show that the impact is significantly higher in industries with greater scope for quality differentiation and for firms with greater absorptive capacity. A different concern about our findings is that they seem puzzling relative to the "imports-as-market-discipline hypothesis" (Levinsohn, 1993). However, this puzzle is only one at first sight as the hypothesis predicts a negative effect of import competition on price-cost margins, hence focusing on the mark-up component of prices and not on the prices themselves. Using the methodology proposed by Roeger (1995), we explicitly test whether our main result contradicts the imports-as-market-discipline hypothesis. We find no contradiction as our results show an insignificant impact of import competition on Chilean firms' price-cost margins. Hence the estimated changes in firms' prices as a result of import competition are not related to changes in firms' mark-ups. Finally, our main finding is robust to a variety of tests on the validity of transport costs as an exogenous barrier to trade, the use of alternative outlier criteria, and the inclusion of additional or alternative control variables.

Our hypothesis of the mechanism driving our results is that import competition forces firms to innovate by offering better-quality products. We test for the validity of this hypothesis by considering alternative interpretations for our results. First, our results could be picking up

the impact of access to better imported intermediate inputs on product quality. Accounting explicitly for the effects of intermediate inputs' transport costs, we show that the importance of stronger import competition for quality upgrading is maintained, even if the access to better inputs also has a significant positive effect. Second, our main findings could be explained by a strategic product market positioning story consisting of a decline in the quality of Chilean imports due to reductions in transport costs (as predicted by Hummels and Skiba (2004)) and Chilean firms improving the quality of their products to fill out the product space (as predicted by Vogel (2008)). However, the story is not supported by our data as the quality of Chilean imports (proxied by their unit values) did not change with transport costs in a manner consistent with the Hummels and Skiba (2004) prediction. Third, our results could reflect an increase in access to export markets if reductions in transport costs for Chilean imports are correlated with decreases in transport costs for Chilean exports. However, our evidence does not support this theoretically plausible explanation as the effects of import competition on product quality are stronger for firms that do not export and for products that are not exported.

Our paper relates to the vast theoretical and empirical literature on the impact of product market competition - of domestic or foreign origin - on firm performance, notably on technological and innovation outcomes.<sup>2</sup> On the theoretical side, Schumpeter (1942) argues in a seminal contribution that producers facing less competition are best placed to innovate since getting adequate returns for one's innovation requires some form of temporary monopoly power. Rodrik (1992) shows that by reducing the firm's market share import competition actually decreases its incentives to innovate. In contrast, stronger competition may foster innovation as producers need to escape their innovating peers to stay in business (Nickell, 1996). Goh (2000), Thoenig and Verdier (2003), Atkeson and Burstein (2007), Ederington and McCalman (2008), and Bustos (2009) show that stronger import competition increases firms' incentives to upgrade their technology and innovate.3 In Thoenig and Verdier (2003) in particular, firms engage in defensive skill-intensive innovations desiring to reduce future threats of imitation or leapfrogging by competitors. Finally, Aghion et al. (2005, 2006) show that the relationship between competition and innovation has an inverted U-shape based on a model which allows for counteracting 'escape competition' effects as well as 'Schumpeterian' effects of competition on innovation depending on firm or industry distance to the technological frontier. While the studies on competition and innovation are not clear-cut about the sign of that relationship, many models suggest that stronger import competition leads to innovation. This is the hypothesis that we will test in the paper focusing on incremental innovation reflected in product quality upgrading.

On the empirical side, studies on the effects of import competition on firm performance or innovative behavior have mostly focused on total factor productivity (TFP) as it captures

<sup>2.</sup> Tybout (2000) surveys the literature on firm adjustments to import competition while Ahn (2002) reviews the literature on competition and innovation.

<sup>3.</sup> Note that the models considered in the various papers are very different: Goh (2000), Thoenig and Verdier (2003), and Rodrik (1992) consider representative firm models whereas Atkeson and Burstein (2007), Ederington and McCalman (2008), and Bustos (2009) consider dynamic heterogeneous firm models.

process innovation. The studies generally find a positive and significant effect of stronger import competition on firm TFP (e.g. Harrison, 1994; Pavcnik, 2002; Fernandes, 2007). Studies examining the effects of import competition on firm product innovation outcomes are rare. Focusing on firms in developed countries, the effect of broad import competition on the involvement in product upgrading or innovation measured by an affirmative answer to the question: 'Did you introduce new or significantly improved goods' is estimated by Bertschek (1995) and Baldwin and Gu (2004). Lelarge and Nefussi (2008) estimate the impact of import competition from low-wage countries on research and development (R&D) spending, while Bloom *et al.* (2009*a*) look at the impact of Chinese import competition on patents. These studies find a positive and significant effect of import competition on innovation outcomes To the best of our knowledge only two studies, Bustos (2009) and Teshima (2009), focus on firms in developing countries but, in contrast to our study, they examine the effects of import competition on a more radical type of innovation, R&D spending.<sup>4</sup>

Our study's contributions to the literature are three-fold. First, ours is the first paper to examine the impact of import competition on a measure of incremental innovation based on firm-product data. In emerging economies where most producers lag behind the world's technology frontier and often improve upon products imported from developed economies, incremental innovation is much more prevalent than the radical R&D-intensive innovation considered in previous studies. We measure incremental innovation using direct quantitative information on firms' product prices rather than subjective perception-based measures of product upgrading as in previous studies. Second, we analyze the effects of import competition on quality upgrading for the universe of Chilean manufacturing products whereas most previous studies focus on the quality of exported products only. Since 86% of the products manufactured by Chilean firms are sold only in the domestic market, this feature of the analysis is important. Furthermore, exported products may differ in many respects from domestically sold products, thus estimates obtained focusing exclusively on the former may be biased. Third, our identification of the effects of import competition on product quality relies on the use of a measure of transport costs that can be considered to be exogenous to quality upgrading.

Our findings indicate that increased exposure to import competition may be beneficial by promoting the innovativeness of producers as a solution to escape competitive pressures from abroad (Pietrobelli and Rabellotti, 2006). Our evidence also suggests that such benefits only arise if the right conditions hold: it requires firms to dispose of skilled personnel and will occur mostly in industries whose attributes offer opportunities for such innovativeness. This gives some support to the idea that the industries in which an economy specializes matter – at least when it comes to benefitting from competition though incremental innovation. While our findings focus on import competition, we believe that they also point to the benefits of competition policy more generally as a central factor to promote innovation. Note that we also find support for another

© OECD 2010 11

\_

<sup>4.</sup> Bustos (2009) uses a measure of expenditures on R&D, computers, software, technology transfers, and patents.

<sup>5.</sup> Without advocating active industrial policies per se, we note that our findings are in line with the conclusions of Hausman et al. (2007) on the importance of a country's specialisation patterns for its development.

explanation for quality upgrading, which is the benefits from easier access to foreign inputs. The latter supports the large literature on learning from embodied foreign technologies as a central channel to stimulate indigenous innovation in emerging economies (e.g. Rivera-Batiz and Romer, 1991; Goldberg *et al.*, 2008).

The remainder of the paper proceeds as follows. Section II describes the data. Section III presents the empirical specification and discusses the use of transport costs as an exogenous proxy for import competition. Section IV discusses our main results, evidence of quality upgrading, and robustness tests. Section V presents alternative explanations for our results. Section VI concludes.

#### II. DATA

#### II.1. Firm Unit Values and Other Information

In our analysis, we use a dataset with information on products at the firm level from 1997 to 2003 that is merged with the annual manufacturing census of Chilean firms with more than 10 employees (ENIA).<sup>6</sup> Both datasets are provided and collected by the Chilean National Statistical Office. The products dataset includes information for each firm and year on the physical quantity sold and the sales value of each of 1 817 products at the 7-digit ISIC level (revision 2). Appendix Table 1 provides some examples of 7-digit ISIC level categories to illustrate the level of detail of the products. The ENIA census described in detail in Fernandes and Paunov (2008) is an unbalanced panel of firms capturing entry and exit that includes information on basic firm characteristics such as employment or ownership, and on accounting variables such as sales.

For each product p7 of firm i in year t we construct a unit value as  $UV_{ii}^{p7} = S_{ii}^{p7}/Q_{ii}^{p7}$ , where S is the value of sales and Q is the physical quantity sold. A unit value measures the average price charged by a firm for each product in a year. Our dataset reports the physical quantities of the 1817 products in 20 different measurement units, some of which are shown in Appendix Table 1. The unit values of products measured in different units (e.g. kilogram, litre) are not comparable. To obtain our final estimating sample, we address two measurement unitrelated issues: i) a few firms do not report the measurement unit of their products' quantity, and ii) some firms report their products' quantity in a different unit than the unit in which the majority of firms report product quantities. The unit values of both types of firms cannot be compared to those of other firms producing the same 7-digit product and are thus excluded from the final sample. Further, to eliminate potential outliers we exclude observations with unit values above (below) the 75th (25th) percentile plus (minus) by 1.5 times the inter-quartile range of the distribution of unit values for any 7-digit product.7 Appendix 2 describes the cleaning procedures and some tests performed to assess the goodness of the products dataset. Our final sample combining the products dataset with the ENIA census includes 51 349 firm-year-product observations corresponding to 5 705 firms with the average number of products manufactured per firm being 2.1. Navarro (2008) shows that many stylised facts based on the Chilean products

<sup>6.</sup> Note that the ENIA dataset provides information by plant and not by firm. However, according to Pavcnik (2002) more than 90% of firms during the 1979-1986 period are single-plant firms. Hence plant information corresponds to a large extent to firm information and we use the term firm throughout the paper.

<sup>7.</sup> While we base our main results on the exclusion of outliers for product categories, our main findings are maintained when the exclusion of outliers is done for product-year categories.

dataset are similar to those obtained for a US products dataset by Bernard *et al.* (2009*a*) and an Indian products dataset by Goldberg *et al.* (2009).<sup>8</sup>

The maintained assumption in our paper is that an increase in unit values proxies for firm product quality upgrading. Table 1 shows average coefficients of variation in unit values for selected 4-digit industries. They show a substantial degree of heterogeneity in unit values across firms and point to some interesting differences across industries. Industries with homogeneous products and thus less scope for quality differences such as cement or petroleum refineries are characterised by low average coefficients of variation. In contrast, industries where quality is expected to play an important role such as electrical machinery, motorcycles, and professional equipment are characterised by higher coefficients of variation. These statistics seem to support our maintained assumption that unit values proxy for the quality of Chilean manufactured products. In Section IV.3 we will provide econometric evidence in support of that assumption.

Table 1: Heterogeneity in Unit Values within Selected 4-digit Industries

4-digit ISIC		Coefficient of Variation
3111	Slaughtering, preparing and preserving meat	4.6%
3114	Canning, preserving and processing of fish, crustaces and similar	9.7%
3134	Soft drinks and carbonated waters industries	5.0%
3212	Manufacture of made-up textile goods	51.4%
3220	Manufacture of wearing apparel	81.9%
3312	Manufacture of wooden and can containers	37.2%
3420	Printing, publishing and allied industries	30.3%
3530	Petroleum refineries	5.5%
3620	Manufacture of glass and glass products	47.9%
3610	Manufacture of pottery, china and earthenware	22.9%
3692	Manufacture of cement, lime and plaster	7.4%
3831	Manufacture of electrical industrial machinery	34.6%
3844	Manufacture of motorcycles and bicycles	70.5%
3851	Manufacture of professional and scientific, and measuring and	86.6%
	controlling equipment n.e.c.	00.0%
3901	Manufacture of jewellery and related articles	28.2%

Notes: The table shows for each 4-digit industry the simple average across all sample years of the industry's yearly coefficients of variation in unit values. For each 4-digit industry and year, the yearly coefficient of variation in unit values is obtained as a weighted average of the coefficients of variation in unit values for each of its 7-digit products using as weights the share of each 7-digit product in the 4-digit industry's total sales in the year.

## **II.2.** Transport Costs

Our measure of transport costs is based on detailed information provided by the Latin American Integration Association (ALADI) on freight costs excluding insurance costs and the free on board customs value (fob) of Chilean imports for each 8-digit Harmonized System (HS)

<sup>8.</sup> For example, the average shares of the most important product, the second most important product, and so on, in total sales of Chilean multi-product plants are strikingly similar to those of US and Indian multi-product plants.

1996 nomenclature code, exporting country, and year from 1997 to 2003. First, we compute for 8-digit HS code m from exporting country c in year t ad-valorem freight rates as the ratio of freight costs ( $freight_{mct}$ ) to the fob value of Chilean imports ( $fob_{mct}$ ):  $fr_{mct} = freight_{mct} / fob_{mct}$ . Second, we aggregate these freight rates from the 8-digit HS code, exporting country, and year level to the 4-digit ISIC (revision 2) and year level using: i) a concordance between 8-digit HS and 4-digit ISIC codes and ii) weights given by Chile's 8-digit HS fob imports from each exporting country and year as a ratio to Chile's total imports in the corresponding 4-digit ISIC code in that year. Appendix 2 provides more details on the construction of the transport costs measure  $TC_t^{k4}$ . We discuss at length the appropriateness of transport costs as an exogenous proxy for import

competition across industries in Section III.2.

<sup>9.</sup> Since some countries may not export a product to Chile due to prohibitive transport costs, our measure is a lower bound for transport costs accounting only for those of exports that actually occur (Hummels, 2001). However, as this feature of our measure is common to products in all industries, it does not impair our analysis which focuses on differences in the relative, rather than the absolute, magnitude of transport costs across 4-digit industries and time.

#### III. EMPIRICAL FRAMEWORK

#### III.1. Baseline Specification

Using the unit values and transport costs measures as defined above we propose the following specification to examine the impact of import competition on product quality:

$$\log UV_{it}^{p7} = \beta * TC_{t-1}^{k4} + \gamma * X_{it} + f_i * I^{p7} + I^{m3} * I_t + \varepsilon_{it}^{p7}$$
(1)

where  $\log UV_{it}^{p7}$  is the log of the unit value for 7-digit product p7 manufactured by firm i in year t,  $^{TC_{t-1}^{k4}}$  are transport costs for 4-digit industry k4 to which the firm's product p7 belongs in year t-1,  $^{X_{it}}$  is a vector of controls to be specified below,  $^{f_{i}} *^{I^{p7}}$  are firm-7-digit product fixed effects,  $I^{m3} *^{I_{t}}$  are 3-digit industry m3-year fixed effects, and  $^{\mathcal{E}_{it}^{p7}}$  is an independent and identically distributed (i.i.d.) residual. The hypothesis that we test in the paper is whether  $^{\beta}$  is negative which would indicate a positive impact of import competition on product quality.

# III.2. Transport Costs: An Exogenous Proxy for Import Competition

# III.2.1. A Proxy for Import Competition

Our paper is not the first to use imports' transport costs as a proxy for import competition: Bernard *et al.* (2006*a*) use transport (combined with tariffs) in a study of the responses of US manufacturing industries to stronger import competition. Notwithstanding, it is important to discuss the reasons that support the idea that transport costs are an adequate proxy for import competition in general and in the case of Chile during our sample period. First, transport costs are an important friction to international trade and due to their size and variability play an important role in shaping patterns of trade across goods and partners.<sup>10</sup> Transport costs represent currently a greater share of total trade costs than the tariffs for most countries due to the trade liberalisation efforts of the last decades (Anderson and Wincoop, 2004). Second, since the 1980s' trade liberalisation, Chile reduced tariffs significantly and

<sup>10.</sup> Hummels (2001) shows that within disaggregate product categories, exporters with the lowest freight rates have the largest import shares based on data for the United States, New Zealand, Argentina, Brazil, Chile, Paraguay and Uruguay.

imposed a uniform tariff structure across industries.<sup>11</sup> Hence, transport costs, rather than tariffs, capture the differential obstacles to trade faced by Chilean industries (Moreira and Blyde, 2006). More importantly, we show that for Chilean industries during our sample period reductions in transport costs led indeed to increases in import competition. Table 2 presents the results from estimating the specification  $IMP_t^{k4} = \alpha_{TC} *TC_{t-1}^{k4} + I_t + I^{m3} + u_t^{k4}$ , where  $IMP_t^{k4}$  is the import penetration ratio of 4-digit industry k4 in year t,  $TC_{t-1}^{k4}$  are transport costs,  $I_t$  are year fixed effects,  $I^{m3}$  are 3-digit industry fixed effects, and  $u_{it}^{k4}$  is an i.i.d. residual.<sup>12</sup>

Dependent Variable: 4-Digit Industry Import Penetration Ratio IV IV (2) (3) -0.571\*\*\* Transport Costs t-1 -0.970\*\*\* -1.377\*\* (0.211)(0.285)(0.629)Year Fixed Effects Yes Yes Yes 3-Digit Industry Fixed Effects Yes Yes Fuel Fuel Instrument price\*transport price\*distance\*import cost first year share first year Number of Observations 465 462 462 0.72 R-Squared

Table 2: Effects of Transport Costs on Import Penetration

Notes on IV-Regression reported in (6): Transport costs are instrumented using measures explained further in the appendix. Results on the underidentification test (Kleinberg LM statistic) are 17.10 (with a P-Value of 0.00) for IV-regressions reported in column (2) and 6.845 (with a P-Value of 0.01) for those in column (3). The F-test of excluded instruments is 31.72 (with a P-Value of 0.00) and 9.88 (with a P-value of 0.00) for columns 2 and 3 respectively.

The results from OLS estimation in column (1) show that as transport costs decline, import penetration ratios increase significantly. An expansion in an industry's imports may reduce its transport costs if the transportation of larger shipments benefits from economies of scale.<sup>13</sup> To mitigate this potential reverse causality, we present in columns (2)-(3) the results from instrumental variables (IV) estimation where transport costs are instrumented by fuel prices – that capture important time-varying shocks to the cost of transportation – interacted with

<sup>11.</sup> Chile's uniform tariff continued to decline from 11% in 1998 to 6% in 2003 but with no variation across industries. Chile signed important trade agreements with the EU and the US (Chumacero et al., 2004). However, those entered into force only after the end of our sample period.

<sup>12.</sup> The import penetration ratio is given by the ratio between imports and the sum of total imports and total domestic sales in the industry as detailed in Appendix 2.

<sup>13.</sup> Hummels (2007) argues that scale economies led to important reductions in shipping prices over the last decades. Indeed, simple correlations between our disaggregate freight rates and the quantity exported for products with more than 30 exporting countries are negative and significant providing suggestive evidence of the presence of economies of scale in transportation. These results make use of quantity data from COMTRADE described further below.

variables allowing those shocks to differentially affect products depending on their transportation intensity. <sup>14</sup> The IV results show a negative and significant effect of transport costs on import penetration. We can therefore argue with confidence that transport costs are an adequate proxy for import penetration across Chilean 4-digit industries.

## III.2.2. Endogeneity Issues

Our imports' transport costs measure is computed based on freight costs that exclude insurance costs. This is an advantage relative to the measure used by Bernard *et al.* (2006*a*) as it avoids one source of endogeneity: the fact that insurance costs increase with the value – and likely with the quality – of exported products. Since our transport costs measures capture the costs incurred by imports from the exporting country until arrival to the point of entry into Chile, they are immune to political economy forces that could pressure for increased trade openness and/or investments in domestic infrastructure as a result of domestic quality improvements. This is an advantage of our measure over tariff-based analyses of the effects of import competition.

While the arguments above suggest that transport costs are exogenous to quality upgrading in Chile, we pursue this issue further given that our analysis does not rely on a radical trade liberalisation episode but rather on a gradual and continuing process of trade integration captured by the variability of transport costs across 4-digit industries and over time. Table 3 illustrates that variability.<sup>15</sup> It is important to show that the origins of this variability are exogenous to product quality upgrading in Chile. The variability in transport costs can originate

in variability of the underlying disaggregate freight rates  ${}^{TC}_{ict}$  across exporting countries and 8-digit HS products. This latter variability can be explained by factors such as infrastructure quality in the exporting country, the distance shipped, the mode of transportation and its quality, market power in the shipping industry, and the type of product being shipped (e.g. heavy versus light) (Limão and Venables, 2001; Hummels, 2007; Hummels *et al.*, 2009). All these factors can be confidently considered to be exogenous to Chilean products' unit values.

18 © OECD 2010

\_

<sup>14.</sup> The variables are transport costs in the first sample year or distance weighted by the import share in the first sample year, described in detail in Appendix 2.

<sup>15.</sup> For certain industries transport costs increase in parts of the sample period which could be due to increases in fuel costs, port congestion, and their differential effect across countries and products despite technological progress.

<sup>16.</sup> The coefficient of variation of the ratio of  $fr_{mct}$  to the average of  $fr_{mct}$  by 8-digit HS product and year is 0.99 and that of the ratio of  $fr_{mct}$  to the average of  $fr_{mct}$  by exporting country and year is 0.64.

Table 3: Transport Costs for Selected 4-digit Industries and Years

4-digit ISIC		1997	1999	2002
3112	Manufacture of dairy products	7.98%	6.46%	6.25%
3118	Sugar factories and refineries	10.67%	15.67%	14.02%
3212	Manufacture of made-up textile goods except wearing apparel	6.69%	7.74%	8.60%
3220	Manufacture of wearing apparel except footwear	4.98%	5.35%	5.13%
3312	Manufacture of wooden and cane containers and small cane ware	9.15%	6.29%	6.11%
3320	Manufacture of furniture and fixtures, except primarily of metal	13.72%	12.25%	13.98%
3122	Manufacture of prepared animal feeds	15.61%	12.91%	12.74%
3133	Malt liquors and malt	19.49%	12.61%	15.66%
3140	Tobacco manufactures	8.19%	8.46%	8.79%
3215	Cordage, rope and twine industries	4.33%	5.08%	6.39%
3233	Manufacture of leather and leather substitutes, except footware and wearing apparel	8.29%	9.85%	9.06%
3240	Manufacture of footwear, except vulcanised or moulded rubber and plastic footwear	5.20%	5.50%	5.81%
3412	Manufacture of containers and boxes of paper and paperboard	15.14%	10.52%	10.41%
3512	Manufacture of fertlizers and pesticides	11.21%	11.91%	10.95%
3551	Tyre and tube industries	7.95%	7.69%	8.25%
3560	Manufacture of plastic products not elsewhere specified	10.27%	10.04%	9.13%
3620	Manufacture of glass and glass products	13.49%	14.31%	13.84%
3720	Non-ferrous metal basic industries	4.64%	4.58%	4.06%
3822	Manufacture of agricultural machinery and equipment	6.51%	5.36%	6.21%
3831	Manufacture of electrical industrial machinery and apparatus	4.85%	4.63%	4.80%
3852	Manufacture of photographic and optical goods	3.36%	3.36%	3.85%
3420	Printing, publishing and allied industries	8.04%	8.74%	8.24%
3522	Manufacture of drugs and medicines	3.30%	3.07%	3.31%
3610	Manufacture of pottery, china and earthenware	11.85%	15.67%	13.97%
3710	Iron and steel basic industries	10.59%	10.06%	10.15%
3812	Manufacture of furniture and fixtures primarily of metal	12.20%	11.35%	12.94%
3813	Manufacture of structural metal products	9.80%	7.65%	8.05%
3844	Manufacture of motorcycles and bicycles	8.69%	10.56%	11.49%

Note: The table shows for each 4-digit industry transport costs aggregated from the level of the 8-digit HS code, exporting country, and year to the level of the 4-digit ISIC and year using as weights the share of each 8-digit HS code and country in Chile's fob imports from that 4-digit ISIC industry in that year.

However, the variability in transport costs might also originate in compositional effects given the way in which the measure is constructed, described in Section II.2. A change in the composition of exporters or products exported to Chile changes the 4-digit transport costs measure even if all disaggregate freight rates are unchanged. This would be problematic if the change in exporters or products responded directly to quality upgrading in Chile.<sup>17</sup> Our data

<sup>17.</sup> One possible reverse causality argument is that domestic improvements in product quality could motivate certain countries to stop exporting or to export smaller volumes to Chile. This would bias downward our estimated effect of transport costs on unit values – making it appear smaller than what it might be in reality – if these changes led to a reduction in our transport costs measure. By construction our transport costs measure would decline if *i*) the countries that stop exporting had relatively high freight rates and represented a large share of imports in the industry, and/or *ii*) the

shows that among the relationships that could impact more substantially the transport costs measure – Chile's most important import relationships – very few start or stop during the sample period.<sup>18</sup> We test further the importance of compositional effects in Section IV.4 by using transport costs measures based on the freight rates of the top exporters to Chile and show that they do not drive our results. Moreover, domestic improvements in product quality could lead countries to increase the quality of their exports to Chile to keep up with local competition. Such an increase proxied by an increase in Chilean imports' unit values – i.e. the ratio of imports fob to the quantity imported - would result from imports fob increasing more than the quantity imported. Since imports fob are the denominator of our disaggregated freight rates, our transport costs measure would decline. Such decline due to an increase in Chilean imports' quality could bias downward our estimated effect of transport costs on unit values - making it appear smaller than what it might be in reality. To examine the validity of this concern, we compute changes in Chilean imports' quality during our sample period using unit values at the 4-digit ISIC and exporting country level based on data from COMTRADE.<sup>19</sup> The median and average annual changes in imports' unit values are 0% and 0.9%, respectively. Hence, the quality of imports to Chile does not seem to have changed substantially over the sample period to sustain concerns of reverse causality.<sup>20</sup> Notwithstanding, we approach this concern further in Section IV.4 by considering an alternative measure of transport costs per unit. This measure avoids the problem that by construction a potential increase in Chilean imports' quality could lead to a decline in our ad valorem transport costs measure.

To mitigate any remaining potential endogeneity concerns and allow the effect of import competition to occur with a lag, we consider the one-year lag of the variable TC as shown in Equation (1) following Bernard *et al.* (2006*a*).

countries sending smaller volumes faced as a result lower freight rates and represented a substantial share of imports in the industry (that would remain large even as exports were reduced). As discussed in the text there is no evidence of *i*). Regarding *ii*), if economies of scale in transportation are at work, a reduction in shipment size would very likely be associated with higher not lower freight rates. Hence this reverse causality argument would if anything result in an increase in our transport costs measure and thus bias our estimates upward making it more difficult to obtain a negative effect of transport costs on quality.

- 18. Considering all of Chile's import relationships at the country-4-digit industry level, 4 960 pairs, only 49% last the entire sample period. Focusing on the top 10 exporting countries to Chile for each 4-digit industry, 4 400 out of 4 764 observations (94%) correspond to relationships that last the entire sample period.
- 19. Unit values are computed as the ratio of Chilean imports fob to the quantity imported from COMTRADE for each 6-digit HS product and exporting country between 1997 and 2003. The COMTRADE data on imports fob is similar to the ALADI data used to construct our transport costs measure summed up from the 8-digit to the 6-digit HS level.
- 20. Note that conceptually it would be equally possible that countries start exporting lower quality products to Chile judging the market there to be too tough on the higher-quality end. That would result in an increase in our transport costs measure which would not be of great concern as it would simply make it more difficult to estimate a negative effect of transport costs on quality. In any case, our evidence on Chilean imports' unit values shows that this possibility did not materialize.

#### III.2.3. Appropriate Level of Disaggregation

Transport costs in Equation (1) are measured at the 4-digit ISIC level. A measure at a more aggregate level may not adequately capture the degree of import competition faced by firms. For example, 3-digit ISIC industry 311, food manufacturing, includes 4-digit industries ranging from fruit and vegetable canning to bakery. If we considered a transport costs measure at the 3-digit level, a reduction in the transport costs of imported bakery products would erroneously suggest that fruit and vegetable canning products also faced stronger import competition, when such products are not exactly substitutes. Certainly, one could argue that measuring transport costs at the 4-digit level for bakery products (ISIC 3117) is still too aggregate. At that level, the measure implies that a reduction in the transport costs of cookie products strengthen the competition faced by cake products too. Cake products may indeed be challenged by imports of cookie products because consumers may decide to substitute cake for cookie products. If import competition was measured at a more disaggregate level – i.e. distinguishing cake from cookie products – then one might wrongly ignore that cross-effect.

Importantly, using a measure at the 4-digit level also reduces potential concerns of a spurious correlation between reductions in transport costs and quality upgrading in Chile due to demand shocks. Suppose that pencils are a 7-digit ISIC product manufactured by Chilean firms and that a positive shock to the demand for pencils increases their unit values domestically. Suppose also that transport costs were measured at the 7-digit level. As response to the demand shock foreign producers could react by exporting more pencils to Chile, this would likely reduce the transport costs on pencils (either because freight costs themselves decline due to scale economies or because imports fob increase). This would result in transport costs and quality (both at the 7-digit level) being spuriously correlated due to the demand shock. The consideration of a transport cost measure at the 4-digit level rather than at the more disaggregated 7-digit level helps to mitigate this type of demand-driven spurious correlation problem. This constitutes a central reason for our choice of using a transport costs measure at the 4-digit level, in addition to the fact that the 4-digit level allows for a reasonable degree of substitutability across products.

#### III.3. Additional Remarks on the Empirical Specification

We now discuss briefly other issues associated with the estimation of Equation (1). First, our specification needs to account for the fact that 49% of Chilean firms manufacture multiple 7-digit products. Among these multi-product firms in any given year, 55% manufacture products within a single 4-digit industry whereas the rest manufacture products across at least two different 4-digit industries.<sup>21</sup> Our specification allows firms manufacturing 7-digit products in various 4-digit industries to face a different degree of import competition – through different transport costs – in each of the 4-digit industries to which their products belong.

Second, Equation (1) controls for crucial firm-product fixed effects. Firms differ in the diversity of products they manufacture, the ways in which they manufacture them, and

<sup>21.</sup> Thus, in any given year about 78% of Chilean plants manufacture products within a single 4-digit industry.

management quality which affects their incentives and possibilities for quality upgrading, maybe differently across products. Given the presence of multi-product firms in the sample, such firm-product fixed effects, rather than firm fixed effects, are what accounts for unobservable firm-product heterogeneity. Their use implies that our main coefficient of interest  $\beta$  is identified on the basis of within-firm changes in the unit value of a given product as transport costs change.

Third, since unit values are prices, they reflect a combination of quality and cost attributes such as input prices and their increase may reflect to some extent an increase in firm market power. Higher costs of production at the firm level may, depending on the degree of competition in the market, lead to increases in unit values unrelated to quality improvements. Controlling for costs is thus important for our main specification and we include the following proxies for production costs: average wages paid by the firm, the share of skilled labor in the firm's total workforce, unit prices paid for electricity by the firm, and the share of imported material inputs in total firm material inputs, further detailed in Appendix 2. Firm size may play a role for quality upgrading by allowing the corresponding fixed costs to be spread over a larger scale and granting easier access to the financing necessary for upgrading, mimicking the role that size plays for radical innovation (Cohen and Klepper, 1996). The vector of controls thus includes a measure of the firm's market share in each of the 4-digit industries to which its products belong and three size dummies based on the firm's total employment, as described in Appendix 2. The downside to including both firm cost as well as market power controls is that doing so raises some endogeneity concerns. For this reason we will examine whether our results hold when these controls are excluded and we will show that this is indeed the case.

Fourth, omitted variables at the industry level correlated with transport costs but also with product quality could bias the estimate of  $^{\beta}$ . The knowledge spillovers generated by FDI in an industry could drive firms, particularly those domestic-owned, to upgrade product quality. However, higher FDI in an industry could also have a negative effect on quality for domestic-owned firms through market-stealing effects. If stronger domestic competition in an industry has 'escape' effects as in Aghion *et al.* (2005), then it would likely be associated with quality upgrading in that industry. To control for these possibilities, we include in the vector of controls the share of total employment in the firm's main 4-digit industry accounted for by foreign-owned firms and the Herfindahl index for each of the 4-digit industries to which the firm's products belong. Bernard *et al.* (2006b) show that multi-product firms can adjust their product range and thus respond differently to increased trade openness relative to single-product firms. We include in the vector of controls a dummy identifying multi-product firms to account for potential differences in their unit values relative to those of single-product firms.

Finally, technological progress, inflation, or other shocks experienced by Chilean industries are accounted for by including 3-digit industry-year fixed effects. These may in particular account for different trends in the prices of materials or capital goods faced by firms in different 3-digit industries which could affect the prices at which they sell their final products.

We believe that Equation (1) allows us to identify an unbiased effect of increased import competition on product quality upgrading at the firm level due to the exogenous nature of transport costs and the set of control variables and fixed effects included.<sup>22</sup>

<sup>22.</sup> Active innovation promotion programmes may affect firms' incentives and possibilities to engage in quality upgrading. However, our specification would need to account for such programmes *only* if they targeted specific industries and could therefore be systematically correlated with transport costs. The Chilean National Fund for Technological and Productive Development (FONTEC) – a public programme in place since 1991 – helped finance innovation projects for manufacturing firms (Benavente *et al.*, 2007). However, the programme did not target specific industries within the manufacturing sector.

#### **IV. RESULTS**

#### IV.1. Main Results

Table 4 presents the results from estimating Equation (1) showing two types of Huber-White standard errors robust to heteroskedasticity. Since our regressions explain firm-product unit values with more aggregate 4-digit transport costs measures, we follow Moulton (1990) and use standard errors clustered by 4-digit industry and year. But we also allow inference to be done based on standard errors clustered by 4-digit industry that allow for serial correlation within industries.<sup>23</sup> All specifications include firm-product fixed effects and 3-digit industry-year fixed effects.

**Table 4: Effects of Transport Costs on Unit Values** 

	Dependent Variable: Log of Firm-Product Unit Value						
	(1)	(2)	(3)	(4)	(5)		
Transport Costs t-1	-1.749**	-1.743**	-1.702**	-1.734**	-1.689**		
	(0.677)	(0.679)	(0.686)	(0.673)	(0.681)		
	[0.806]	[0.809]	[0.812]	[0.798]	[0.804]		
Firm Cost Controls	No	No	No	Yes	Yes		
Other Firm Controls	No	Yes	No	No	Yes		
Industry Controls	No	No	Yes	No	Yes		
Firm*Product Fixed Effects	Yes	Yes	Yes	Yes	Yes		
3-Digit Industry*Year Fixed Effects	Yes	Yes	Yes	Yes	Yes		
Number of Observations	41979	41979	41966	41938	41929		
R-Squared	0.97	0.97	0.97	0.97	0.97		

Notes: Robust standard errors clustered by 4-digit industry and year in round parentheses and clustered by 4-digit industry in square parentheses. \*\*\*, \*\*, and \* indicate significance at 1%, 5%, and 10% confidence levels, respectively. Firm cost controls include the log of average wages, the share of skilled labor in total labor, the log of unit electricity prices paid by the firm, and the share of imported inputs in total inputs. Other firm controls include size dummies, a dummy for multi-product firms, and the firm's market share at the 4-digit level. Industry controls include the share of employment in foreign-owned firms in total 4-digit industry employment and the normalised Herfindahl index at the 4-digit industry level.

The estimates in column (1) show that import competition has a positive effect on product quality when firm cost controls, other firm characteristics, and industry characteristics are

<sup>23.</sup> Our results are also robust to the consideration of standard errors clustered by product-year and by firm-year. In the tables that follow Table 4 we show for simplicity only the standard errors clustered by 4-digit industry and year. However all results are robust to the consideration of standard errors clustered by 4-digit industry.

ignored. In column (2), the specification includes only firm characteristics in addition to transport costs. The estimate of  $\beta$  is negative and significant and its magnitude is unchanged. Columns (3) and (4) show the results from specifications where in addition to transport costs either only industry characteristics or only firm cost controls are included, respectively. The estimates of  $\beta$  are again maintained. Column (5) shows our preferred specification which includes the three types of controls. The fact that our preferred results do not change if firm cost and market power controls are excluded, reassures us that the potential endogeneity concerns associated with those controls are not driving our results. Also, note that the control variables are contemporaneous relative to firm unit values but qualitatively similar results are obtained when one-year lagged control variables are considered.

Given that unit values are in logarithms and transport costs are measured in fractional terms, the estimate of  $\beta$  implies that a 3 percentage point reduction (slightly less than a one standard deviation) in transport costs would lead to an average increase in unit values of almost 5.2% within firms and products. Such reduction in transport costs would be associated with the following increases in actual unit values: e.g. i) from an average of USD 86 to USD 108 for bicycles; ii) from an average of USD 227 to USD 299 for domestic ovens and iii) from an average of USD 16 454 to USD 26 996 for fabricated motor vehicles.<sup>24</sup>

While Section III.2 shows that transport costs are an adequate proxy for import competition, we also estimate Equation (1) substituting import penetration ratios of 4-digit industries for transport costs. Then, a positive  $^{\beta}$  indicates that stronger import competition stimulates firm quality upgrading. Table 5 presents the corresponding results where again all specifications include firm-product and 3-digit industry-year fixed effects.

© OECD 2010 25

\_

<sup>24.</sup> These averages are for the year 2000 and the unit values are expressed in USD using the corresponding average peso-USD exchange rate obtained from the Central Bank of Chile. Providing an economic magnitude for the average product is not possible due to the lack of comparability of unit values across products measured in different units.

Table 5: Effects of Import Penetration on Unit Values

	Dependent Variable: Log of Firm-Product Unit Value						
						IV-Reg	
	(1)	(2)	(3)	(4)	(5)	(6)	
Import Penetration t	0.211**	0.201**	0.199**	0.209**	0.185**	1.706*	
	(0.089)	(0.089)	(0.090)	(0.089)	(0.089)	(0.962)	
Firm Cost Controls	No	No	No	Yes	Yes	Yes	
Other Firm Controls	No	Yes	No	No	Yes	Yes	
Industry Controls	No	No	Yes	No	Yes	Yes	
Firm*Product Fixed Effects	Yes	Yes	Yes	Yes	Yes	Yes	
3-Digit Industry*Year Fixed Effects	Yes	Yes	Yes	Yes	Yes	Yes	
Number of Observations	50797	50797	50787	50764	50754	43590	
R-Squared	0.96	0.96	0.96	0.96	0.96		

Notes: Robust standard errors clustered by 4-digit industry and year in parentheses in columns (1)-(5) and clustered by 4-digit industry in column (6). \*\*\*, \*\*, and \* indicate significance at 1%, 5%, and 10% confidence levels, respectively. Firm cost controls include the log of average wages, the share of skilled labor in total labor, the log of unit electricity prices paid by the firm, and the share of imported inputs in total inputs. Other firm controls include size dummies, a dummy for multi-product firms, and the firm's market share at the 4-digit level. Industry controls include the share of employment in foreign-owned firms in total 4-digit industry employment and the normalised Herfindahl index at the 4-digit industry level. In column (6) import penetration is instrumented using our lagged transport costs measure and industry-specific exchange rates described in Appendix 2. The p-value of the corresponding F-test for excluded instruments is 0.03, the p-value of the corresponding over-identification test (Hansen J-statistic) is 0.24 and the p-value of the under-identification test (Kleibergen-Paap LM statistic) is 0.06. The standard errors in column (6) are clustered by 4-digit industry rather than by 4-digit industry and year since the latter is not feasible due to a lack of degrees of freedom.

Columns (1)-(5) reproduce the specifications in columns (1)-(5) of Table 4, i.e. progressively adding firm and industry controls. The estimates show that import competition has a positive and significant effect on product quality. However, import penetration ratios may be endogenous as the decisions of foreign exporters are likely to be affected by Chilean firms' product innovation choices. Hence, we present in column (6) the results from an IV specification where import penetration ratios are instrumented by lagged transport costs and industry-specific exchange rates described in Appendix 2. The results show a positive effect of import penetration on product quality that is significant at a 10% confidence level.

#### IV.2. Unit Values and Quality

The maintained assumption in this paper is that increases in unit values are a proxy for product quality upgrading and represent incremental innovation. For certain consumer products such as automobiles or washing machines, it is clear that higher prices are directly correlated with higher quality. This explains why so many studies in the trade literature have taken for granted the idea that increases in export unit values represent improvements in quality. The summary statistics on the heterogeneity in Chilean products' unit values presented in Table 1 support this argument. Industries with little scope for quality differences show low relative variation in unit values while industries where quality is expected to play an important role such as professional equipment (which includes information technology products) exhibit a much higher variability in unit values. The role of prices as a signal for quality has been considered in the industrial organisation literature. In the presence of imperfect information, prices are a good

signal for quality since firms often choose intentionally their level as to reveal to consumers the higher quality of their products. Fluet and Garella (2002) show theoretically that in markets with strong vertical product differentiation (i.e. those with substantial quality differences within product categories) firms may base their signaling on prices only. Thomas *et al.* (1998) provide empirical evidence showing that higher prices are used for quality signaling purposes in the US automobile industry.

We should note, however, that Khandelwal (2009) criticizes the assumption that unit values equal quality by arguing that prices could also reflect variations in manufacturing costs. Based on a model that allows consumers to have preferences for vertical attributes (quality) as well as horizontal attributes (e.g. colour given a fixed quality) of products, he shows that conditional on price (imported) products with higher market share are assigned higher quality or the mapping of prices to quality is not direct but depends on market share (desirability of the product) too. However, he does show that the mapping of unit values to quality is very appropriate in the industries with more scope for quality differentiation.

Given the potential criticisms we now explore whether our assumption that price increases reflect quality improvements following stronger impact competition is indeed justified. Hummels and Klenow (2005) show that richer countries tend to export higher quality products. We start by testing empirically a simple hypothesis related to that finding: stronger import competition due to increased imports from relatively poor countries is unlikely to lead to product quality upgrading as their products have lower quality than those of Chilean firms. This hypothesis is likely to be verified particularly for the reductions in transport costs of exporting countries very much behind Chile's level of economic development such as those with a GDP per capita less than 50% of Chile's. Accordingly, we obtain transport costs measures separately for

less advanced countries  $TC_{it-1}^{k4 \ lessadv}$  and for more advanced countries  $TC_{it-1}^{k4 \ moreadv}$  and we estimate the following specification: <sup>25</sup>

$$\log UV_{it}^{p7} = \overline{\beta}_{TCL} * TC_{it-1}^{k4 \ lessadv} + \overline{\beta}_{TCM} * TC_{it-1}^{k4 \ moreadv} + \gamma * X_{it} + I^{p7} + I^{m3} * I_{t} + f_{i} + \varepsilon_{it}^{p7}$$
(2)

where all variables other than the transport costs are defined as before. The results are reported in column (1) of Table 6 and confirm the simple hypothesis: i.e. only reductions in transport costs of advanced countries lead to quality upgrading and the difference between the effects of transport costs for both groups of countries is significant. One may argue that although these less advanced exporters to Chile do not export high quality products they could still spur innovation there. Bloom *et al.* (2009*a*, 2009*b*) find such evidence for the impacts of Chinese import competition on innovation in Europe and present a model that rationalizes such impacts. We, therefore, take these results as merely suggestive and undertake a series of alternative tests.

© OECD 2010 27

\_

<sup>25.</sup> We define less and more advanced countries based on their GDP per capita in constant 2005 USD in any year relative to Chile's GDP per capita in that year using data from the World Development Indicators. The results are qualitatively similar if the average GDP per capita in constant 2005 USD for the 1997-2003 period is used instead.

Table 6: Effects of Transport Costs on Unit Values – Evidence of Quality Upgrading

		Depend	lent Variable: Log o	f Firm-Product Un	it Value	
	(1)	(2)	(3)	(4)	(5)	(6)
Transport Costs of Less Advanced Countries t-1	-0.014					
	(0.063)					
Transport Costs of More Advanced Countries t-1	0.968*					
	(0.523)					
Transport Costs t-1 * Rauch Differentiated Industries		-2.618**				
		(1.038)				
Transport Costs t-1 * Rauch Non-Differentiated Industries		-1.142*				
		(0.605)				
Transport Costs t-1 * Industries with Larger Advertising Intens	ity		-2.207***			
			(0.716)			
Transport Costs t-1 * Industries with Smaller Advertising Inten	sity		-0.528			
			(0.645)			
Transport Costs t-1 * Industries with Larger Patent Stock				-2.099***		
				(0.681)		
Transport Costs t-1 * Industries with Smaller Patent Stock				0.074		
				(0.453)		
Transport Costs t-1 * Industries with Lower Demand Elasticity					-2.095***	
					(0.628)	
Transport Costs t-1 * Industris with Higher Demand Elasticity					1.310	
					(1.060)	
Transport Costs t-1 * Firms with Higher Skilled Labor Share						-2.642**
						(1.237)
Transport Costs t-1 * Firms with Lower Skilled Labor Share						-0.496
						(0.531)
Firm Cost Controls	Yes	Yes	Yes	Yes	Yes	Yes
Other Firm Controls	Yes	Yes	Yes	Yes	Yes	Yes
Industry Controls	Yes	Yes	Yes	Yes	Yes	Yes
Firm*Product Fixed Effects	Yes	Yes	Yes	Yes	Yes	Yes
3-Digit Industry*Year Fixed Effects	Yes	Yes	Yes	Yes	Yes	Yes
P-value for F-Test of Difference in Coefficients across Groups	0.07	0.12	0.08	0.00	0.01	0.10
Number of Observations	41367	39481	41929	41929	38895	38184
R-Squared	0.97	0.97	0.97	0.97	0.97	0.96

Notes: Robust standard errors clustered by 4-digit industry and year in parentheses. \*\*, and \* indicate significance at 1%, 5%, and 10% confidence levels, respectively. The regressions include our transport costs measure interacted with alternative sets of dummy variables described in the text. The firm cost controls, other firm controls, and industry controls included in the regressions are similar to those in column (5) of Table 4. In column (1), following Rauch (1999), the differentiated goods 4-digit industries are those that are neither: i) homogenous – traded in organised exchanges (e.g. steel) nor ii) reference-priced. In column (2), the industries with larger advertising intensity are those whose advertising intensity is higher than the median intensity across all US industries. In column (3), the industries with larger patent stocks are those whose patent stocks are larger than the median patent stock across all US industries. In column (4), the industries with lower import demand elasticity are those whose import demand elasticity taken from Broda et al. (2006) is below the median across Chilean industries. In column (5), the firms with higher skilled labor share are those whose share of skilled labor in total wages in the first sample year is below the median across Chilean firms.

To provide further support that our estimates refer indeed to product quality, we examine whether the effects of transport costs on unit values are stronger for industries whose product attributes suggest more opportunities for quality improvements or for firms with characteristics likely to be associated with those improvements. We estimate a variant of Equation (1) given by:

$$\log UV_{ii}^{p7} = \beta_1 * TC_{t-1}^{k4} * group1 + \beta_2 * TC_{t-1}^{k4} * group2 + \gamma * X_{it} + f_i * I^{p7} + I^{m3} * I_t + \varepsilon_{ii}^{p7}$$
(3)

where the effect of transport costs is allowed to differ across industries or firms belonging to group 1 and industries or firms belonging to group 2, and all other variables are defined as before.

Column (2) of Table 6 reports the results from estimating Equation (3) considering as group 1 (group 2) differentiated goods industries (non-differentiated goods industries) according to the classification proposed by Rauch (1999).<sup>26</sup> Bastos and Silva (2009) show that the classification scheme is well suited to capture quality differentiation. Our estimates show that the impact of transport costs on quality is larger in magnitude for firms in differentiated goods industries, with the difference in coefficients being significant at slightly more than the 10% confidence level.

Column (3) of Table 6 reports the results from estimating Equation (3) considering as group 1 (group 2) industries with a larger (smaller) advertising intensity. The scope for vertical quality differentiation is assumed to be higher in industries with a 'larger' advertising intensity as in Kugler and Verhoogen (2008). Using data from the 1976 US Federal Trade Commission Line of Business Survey, we compute the ratio of advertising expenditures to total sales for large firms in US industries and use those as a benchmark for our Chilean 4-digit industries.<sup>27</sup> The advertising intensity of an industry is 'larger' if it is higher than the median intensity in the sample. Our findings indicate that the response of quality to import competition is significantly larger for firms in industries with a greater scope for vertical differentiation.

Column (4) of Table 6 shows the results considering as group 1 (group 2) industries with larger (smaller) patent stocks in the US We assume that industries with 'larger' patent stocks in the US present more incremental innovation opportunities for Chilean firms. We use data from the OECD patents database on the number of international patent applications by US 2-digit ISIC revision 3 industries cumulated over time to generate patent stocks.<sup>28</sup> The patent stock of an industry is 'larger' if it is higher than the median patent stock in the sample. Our estimates show that the effect of transport costs on quality is significantly more negative for firms in industries with larger patent stocks, thus more innovation opportunities.

Column (5) of Table 6 reports the results from estimating Equation (2) considering as group 1 (group 2) industries with lower (higher) elasticities of substitution. We use the import

<sup>26.</sup> According to Rauch's classification, differentiated products are those that are neither *i*) homogenous – traded in organised exchanges (e.g. steel) nor *ii*) reference-priced – having listed prices in trade publications (e.g. some chemical products) and require a more important degree of buyer-seller interaction. To use Rauch's classification, we establish a correspondence between his 4-digit SITC rev. 2 codes and our 3-digit ISIC rev. 2 codes. For the printing industry (ISIC 342), we are unable to establish an unambiguous correspondence and thus drop it from the regression.

<sup>27.</sup> We use a concordance provided by Eric Verhoogen to obtain advertising intensities at the ISIC revision 2 4-digit level starting from the FTC 4-digit classification.

<sup>28.</sup> These are patent applications filed under the Patent Cooperation Treaty (PCT) classified according to the international patent classification that is concorded to the 2-digit ISIC revision 3 industry level by Schmoch *et al.* (2003). To obtain patent stocks at the 2-digit ISIC revision 2 level, we use a concordance included in our Chilean census dataset. The patent stock is computed using the perpetual inventory method cumulating annual patents since 1989 and using a depreciation rate of 15%, following that used to compute R&D stocks by Griffith *et al.* (2006).

demand elasticities for Chile estimated by Broda *et al.* (2006).<sup>29</sup> The 'lower' is the elasticity of substitution across imported products within an industry the greater is the extent of product differentiation. The import demand elasticity of an industry is 'lower' if it is below the median elasticity in the sample. The effect of import competition on quality is found to be significantly stronger for firms in industries with lower import demand elasticity, thus more scope for product differentiation.

Human capital is a key component of a firm's absorptive capacity to new technology and knowledge necessary for product quality upgrading (Cohen and Levintahl, 1989). Column (6) of Table 6 shows the results from estimating Equation (2) defining group 1 (group 2) to include firms whose wage share of skilled labor in their first sample year is larger (smaller) than the sample median. The estimates and the F-test show that increased import competition leads to a significantly stronger increase in unit values for firms with larger skill shares.

The findings in Table 6 provide substantial evidence to support our assumption that increases in unit values are a good proxy for product quality improvements. Hence, by focusing on incremental product innovation outcomes, our main findings complement nicely those obtained for radical innovation outcomes as a response to increased import competition from for firms in Argentina and Mexico by Bustos (2009) and Teshima (2009), respectively.

#### IV.3. The Imports-as-Market-Discipline Hypothesis

For a skeptical reader for whom our evidence in Section IV.2 is not sufficient to prove the validity of our assumption that unit values capture the quality of Chilean manufacturing products, our main findings in Table 4 may appear puzzling. Taken literally, our regressions show that Chilean product prices increase with import competition. Yet, the most obvious advantage of more competitive markets is their disciplining effect forcing firms to price closer to marginal costs. Competition through increased imports may be an even more powerful price-reducing mechanism. Trade models with imperfect competition predict that firms' mark-ups are a decreasing function of the elasticity of demand and stronger import competition tends to increase the elasticity of demand (e.g. Helpman and Krugman, 1985; Melitz and Ottaviano, 2008). The prediction of the "imports-as-market-discipline" hypothesis is, therefore, of a negative effect of import competition on mark-ups or price-cost margins. One needs to be cautious when interpreting our results on prices as that hypothesis predicts a reduction in mark-ups not necessarily in prices per se. Hence, before concluding that our results in Table 4 are truly puzzling and at odds with the "imports-as-market-discipline" hypothesis, we need to examine the impact of import competition on Chilean firms' price-cost margins.

Price-cost margins are not observable since marginal costs are not observable; hence we follow the widely used methodology proposed by Roeger (1995) to estimate them. The

<sup>29.</sup> While these elasticities are commonly used to measure the scope of horizontal differentiation, Khandelwal (2009) argues that they are conceptually related to vertical differentiation given the specific estimation method used by Broda et al. (2006). We use a concordance between HS and ISIC to establish a correspondence between their 3-digit HS codes and 4-digit ISIC rev. 3 codes. Those are then matched to our 7-digit ISIC products using a correspondence from 4-digit ISIC rev. 3 codes to ISIC products provided by the INE.

methodology computes the difference between the primal Solow residual in the presence of imperfect competition (Hall, 1988) and the corresponding dual Solow residual derived from a cost function. This difference eliminates firm unobserved productivity which is associated with an endogeneity bias in production function estimation and results in an equation providing consistent estimates for price-cost margins.<sup>30</sup> We allow average price-cost margins to vary with transport costs and with the degree of domestic competition faced by each firm in its main 4-digit industry.<sup>31</sup> Our estimable equation is given by:

$$\Delta Z_{it}^{k4} = \beta_1 \Delta X_{it}^{k4} + \beta_2 \Delta X_{it}^{k4} * TC_{t-1}^{k4} + \beta_3 \Delta X_{it}^{k4} * H_t^{k4} + \delta_1 TC_{t-1}^{k4} + \delta_2 H_t^{k4} + f_i + I_t + \eta_{it}$$
(4)

where  $^{\Delta Z_{it}}$  and  $^{\Delta X_{it}}$  are computed based on the growth of firm nominal sales, wage bill, intermediate costs, and capital as described in Appendix 3,  $^{TC_{t-1}^{k4}}$  is defined as before,  $^{H_t^{k4}}$  is the Herfindahl index in 4-digit industry k4,  $^{I_t}$  are year fixed effects,  $^{f_i}$  are firm fixed effects, and  $^{\eta_{it}}$  is an i.i.d. residual. $^{32}$  The estimate of  $^{\beta_1}$  is the average price-cost margin while those of  $^{\beta_2}$  and  $^{\beta_3}$  show how average price-cost margins change with transport costs and with the degree of domestic competition, respectively. $^{33}$ 

The results from estimating Equation (4) by firm fixed effects are shown in Table 7 with standard errors clustered by 4-digit industry. Columns (1) and (3) show that the average price-cost margins of Chilean firms are negatively related with transport costs. However, the effects are insignificant. In contrast, columns (2)-(3) show that average price-cost margins are positively and significantly linked to domestic competition. Hence, we find no evidence to support or reject the "imports-as-market-discipline" hypothesis: price-cost margins and mark-ups of Chilean firms simply do not vary significantly with import competition. This is not surprising given that the pro-competitive price-lowering effects from imports likely occurred much earlier than our sample period after the radical trade liberalisation of the early 1980s.<sup>34</sup> Note that while many

© OECD 2010 31

\_

<sup>30.</sup> We refer the reader to Roeger (1995) and Konings et al. (2005) for details on the derivation of that equation.

<sup>31.</sup> Firm-level estimates of price-cost margins cannot be obtained due to insufficient degrees of freedom.

<sup>32.</sup> For comparability with the estimates of Equation (1) transport costs are lagged one year. However, the results are qualitatively similar including current transport costs or including all variables lagged one year.

<sup>33.</sup> Note that this analysis uses only firm aggregates in terms of sales and inputs but does not use information at the firm-product level. However, Navarro (2008) shows that Chilean multi-product firms tend to have leading products that represent a very large share of overall production and sales of the firm. Therefore, even for multi-product firms, these results can be considered at least suggestive of the effects on the firm's main product's price-cost margin.

<sup>34.</sup> Indeed, Tybout (1996) studies the impact of radical trade liberalisation on Chilean firms' price-cost margins during the 1979-1986 period and finds significant negative effects on price-cost margins.

empirical studies find negative effects of import competition on price-cost margins not all studies find robust evidence.<sup>35</sup>

Table 7: Effects of Transport Costs on Price-Cost Margins

	Dependent Variable: $\Delta Z_{t}$ (in Equation (4))					
	(1)	(2)	(3)			
ΔX t	0.506***	0.454***	0.492***			
	(0.029)	(0.016)	(0.029)			
ΔX <sub>t</sub> * Transport Costs <sub>t-1</sub>	-0.417		-0.413			
	(0.340)		(0.324)			
$\Delta X_t^*$ Herfindahl Index $t$		0.178**	0.177**			
		(0.084)	(0.081)			
Firm Fixed Effects	Yes	Yes	Yes			
Year Fixed Effects	Yes	Yes	Yes			
Number of Observations	18282	18282	18265			
R-Squared	0.42	0.42	0.42			

Notes: Robust standard errors clustered by 4-digit industry in parentheses. \*\*\*, \*\*\*, and \* indicate significance at 1%, 5%, and 10% confidence levels, respectively. The computation of the dependent variable  $\Delta Z_{it}$  and of  $\Delta X_{it}$  is described in Appendix 3. In columns (1) and (3) the transport costs measure is included in levels and interacted with  $\Delta X_{it}$ .

Stronger import competition has no effect on Chilean firm mark-ups but is linked to average product price increases. The increase in price-cost margins that would result from higher prices must have been counteracted by higher costs incurred by firms. If no meaningful changes to firms' production processes or operations occurred, such explanation would be implausible. If anything, one would expect costs to decrease as firms might, as a result of import increases, access cheaper foreign inputs. However, if higher prices are charged for higher quality products, then the higher costs incurred by firms were necessary to achieve those quality improvements or to signal the quality of their products. This reasoning is in agreement with the firm heterogeneity model proposed by Johnson (2009) where fixed and marginal costs of production increase with the level of product quality chosen by firms. Manova and Zhang (2009) also argue that in response to stronger competition firms may reduce their mark-ups but also increase product quality. If quality upgrading requires fixed costs or more expensive inputs, both marginal and fixed costs may increase and if this effect is sufficiently strong it can dominate any mark-up adjustment and lead to increases in output prices.

<sup>35.</sup> Konings *et al.* (2001) for example find insignificant effects of import competition on the price-cost margins of firms in Belgium. Moreover, the estimated contribution of trade to reducing price-cost margins seems relatively modest (Boulhol, 2009).

#### IV.4. Additional Findings and Robustness

Our main results reported in Table 4 are for the full unbalanced panel containing both the entry and exit of firms as well as the entry and exit of products. It is interesting to investigate how these factors impact on our results. Table 8 shows results for two different sub-samples.

Table 8: Effects of Transport Costs on Unit Values – Extensions

Panel A: Sample of Continued Products

	Dependent Variable: Log of Firm-Product Unit Value					
	(1)	(2)	(3)	(4)	(5)	
Transport Costs <sub>t-1</sub>	-1.955***	-1.971***	-1.904**	-1.944***	-1.914***	
	(0.727)	(0.720)	(0.737)	(0.722)	(0.722)	
Firm Cost Controls	No	Yes	No	No	Yes	
Other Firm Controls	No	No	Yes	No	Yes	
Industry Controls	No	No	No	Yes	Yes	
Firm*Product Fixed Effects	Yes	Yes	Yes	Yes	Yes	
3-Digit Industry*Year Fixed Effects	Yes	Yes	Yes	Yes	Yes	
Number of Observations	18677	18677	18674	18656	18656	
R-Squared	0.96	0.96	0.96	0.96	0.96	

Panel B: Sample of Continuing Firms and Continued Products

Dependent Variable: Log of Firm-Product Unit Value (1)(2)(3)(4) (5)Transport Costs t-1 -2.131\*\*\* -2.113\*\*\* -2.127\*\*\* -2.053\*\*\* -2.031\*\* (0.602)(0.599)(0.591)(0.608)(0.621)Firm Cost Controls Nο Yes Nο Nο Yes Other Firm Controls No No Yes No Yes **Industry Controls** No No No Yes Yes Yes Yes Yes Firm\*Product Fixed Effects Yes Yes 3-Digit Industry\*Year Fixed Effects Yes Yes Yes Yes Number of Observations 12125 12125 12113 12113 12125 0.95 0.95 0.95 0.95 0.95 R-Squared

Notes: Robust standard errors clustered by 4-digit industry and year in parentheses. \*\*\*, \*\*, and \* indicate significance at 1%, 5%, and 10% confidence levels, respectively. Firm cost controls include the log of average wages, the share of skilled labor in total labor, the log of unit electricity prices paid by the firm, and the share of imported inputs in total inputs. Other firm controls include size dummies, a dummy for multi-product firms, and the firm's market share at the 4-digit level. Industry controls include the share of employment in foreign-owned firms in total 4-digit industry employment and the normalised Herfindahl index at the 4-digit industry level. The regressions in Panel A are estimated for the sub-sample of all firms but only products that the firm neither starts producing nor discontinues during its years in the sample while those in Panel B are estimated for the sub-sample of firms included in the sample during the entire sample period and for each of those firms only the products that they produce during the entire sample period.

In Panel A, we use a sub-sample of all firms but only the products that firms neither start producing nor discontinue during their years in the sample (continued products). The effect of transport costs on product quality is significant and more negative than in Table 4. The difference in magnitudes suggests that products with less upgrading potential are likely to be discontinued by firms and new products are also less subject to upgrading as a result of import

competition after their initial introduction. In Panel B, we use a sub-sample including only firms that are in the sample during the entire sample period (continuing firms) including for each of those firms only their continued products. Transport costs have again a significant effect on product quality that is more negative than in Panel B. This difference in magnitudes suggests that the 'well-established' products of continuing firms are more prone to quality upgrading as a response to increased import competition than the continued products of firms which just started operations or those of firms in their years shortly before exit.

While for brevity Table 4 and the tables thereafter do not report the estimated coefficients on the control variables included in our regressions, two findings are noteworthy. Firms with larger market shares exhibit significantly higher unit values. However, this market power effect does not eclipse the importance of increased import competition in generating quality improvements. Also, firms in industries with stronger foreign presence exhibit on average higher unit values. These potential knowledge spillovers from FDI seem to complement – not eliminate – the effects of import competition on quality upgrading.

We conduct an extensive set of tests to verify the robustness of the results from our preferred specification in column (5) of Table 4. First, we address two potential endogeneity concerns related to the effects of compositional changes on our transport costs measure discussed in Section III.2.2. The first concern is the potential change in transport costs if exporters stop supplying the Chilean market. Column (1) of Table 9 shows the results from estimating Equation (1) including a measure of transport costs constructed using only the freight rates of the top 10 exporters to Chile for each 8-digit HS product category in each year. The effect of transport costs on product quality is still negative and significant. Column (2) shows that our results are also maintained if we add an even more restricted measure of transport costs constructed using only the freight rates of the top 3 exporters to Chile (United States, Argentina and Brazil). Since major exporters are less likely to discontinue their products, the results in columns (1)-(2) are assuring that compositional effects of our transport costs measure do not drive our results. The second concern is the potential effect of changes in the quality of imported products. While in Section III.2 we show that there was no major change in that quality, we examine this issue further by considering transport costs per unit. These are obtained based on freight rates per unit defined as:  $fru_{mct} = freight_{mct} / impqt_{mct}$ , where freight costs rates are at the 6-digit HS level and  $impqt_{mct}$  denotes the quantity of HS 6-digit product m imported from country c at time t taken from COMTRADE. Given that all import quantities are measured in tons, we can aggregate these disaggregated freight rates per unit to the 4-digit ISIC and year level as described in Section III.2. An important disadvantage of these transport costs per unit measures is that the COMTRADE database has many missing values for the quantities imported. In order to use a transport costs measure that does reflect the import competition faced by a given 4-digit industry, we construct our transport costs per unit measure only for the 4-digit industries where more than 80% of imports fob in any one year have information on the corresponding quantities. The results from estimating Equation (1) including the log of the transport costs per unit measure are reported in column (3) of Table 9 and show that our main finding is qualitatively maintained.

Table 9: Effects of Transport Costs on Unit Values – Robustness

					Depende	ıt Variable:				
	0,			Estimated Average Firm Unit Value			Log of Firm-Pro	duct Unit Value		
	Alternati	Alternative Transport Costs Measures			Alternative Outlier Criteria for Unit Values		Additional Firm Control		Alternative Competition Measures	
	Include Only Top 10 Exporters by Product-Year	Include Only Top 3 Exporters by Product-Year	Transport Costs per Unit	Second Stage Regression in 2- Stage Procedure	Exclude Top/Bottom 10% of Unit Values by Product	Windsorize Unit Values Based on Quartiles Criterion by Product	Exporter Status	Foreign Ownership Status	Share of Top 5 Firms in 4-digit Industry	Adding Regional Competition Measures
	(1)	(2)	(3)	(4)	(5)	(6)	(7)	(8)	(9)	(10)
Transport Costs t-1	-1.214** (0.578)	-0.545* (0.320)	-0.074* (0.039)	-5.119*** (1.880)	-1.667** (0.801)	-1.449** (0.725)	-1.695** (0.679)	-1.704** (0.675)	-1.705** (0.705)	-1.740*** (0.663)
Firm Cost Controls	Yes	Yes	Yes	No	Yes	Yes	Yes	Yes	Yes	Yes
Other Firm Controls	Yes	Yes	Yes	No	Yes	Yes	Yes	Yes	Yes	Yes
Industry Controls	Yes	Yes	Yes	No	Yes	Yes	Yes	Yes	Yes	Yes
Firm*Product Fixed Effects	Yes	Yes	Yes	No	Yes	Yes	Yes	Yes	Yes	Yes
3-Digit Industry*Year Fixed Effects	Yes	Yes	Yes	No	Yes	Yes	Yes	Yes	Yes	Yes
Number of Observations	41414	41371	24276	19546	36733	44157	41942	41942	41705	39290
R-Squared	0.965	0.965	0.978	0.32	0.969	0.956	0.965	0.965	0.965	0.966

Notes: Robust standard errors clustered by 4-digit industry and year in parentheses. \*\*\*, \*\*, and \* indicate significance at 1%, 5%, and 10% confidence levels, respectively. Firm cost controls include the log of average wages, the share of skilled labor in total labor, the log of unit electricity prices paid by the firm, and the share of imported inputs in total inputs. Other firm controls include size dummies, a dummy for multi-product firms, the firm's market share at the 4-digit level, a dummy for the firm's exporter status (in column (7)) and a dummy for foreign ownership (in column (8)). Industry controls include the share of employment in foreign-owned firms in total 4-digit industry employment and the normalised Herfindahl index at the 4-digit industry level (except in column (9) where the sales share of the largest 5 firms at the 4-digit level is included). In column (10) a regional Herfindahl index at the 4-digit industry-level and regional firm market shares are also included. The transport costs measure used in columns (1) and (2) are computed based only on the freight rates of, respectively, the top 10 and top 3 exporting countries to Chile for each 8-digit HS product in each year. The procedure to obtain the average firm unit values used as dependent variable in column (4) follows Kugler and Verhoogen (2008) and is described in the text. The outlier criteria used in columns (5)-(6) are described in the text.

Second, a possible concern with our estimates is that the coefficient on transport costs is driven by the larger weight of multi-product firms that have by definition more observations per year than single-product firms. To address this possibility, we follow the two-stage regression procedure proposed by Kugler and Verhoogen (2008). First, we regress firm unit values (the dependent variable in Equation (1)) on firm-year, product-year, and year fixed effects. For any given year, the estimated firm-year fixed effect provides an average firm unit value identified by the differences between a firm's unit value(s) and those of other firms producing the same product(s) in that year. Second, these time-varying estimated average firm unit values are

regressed on our transport cost measures along with 3-digit industry-year fixed effects.<sup>36</sup> In this regression a single-product firm and a multi-product firm included in the sample during the same number of years have equal weight. Column (4) of Table 9 presents the results from this regression which indicates that our main finding is qualitatively maintained.

Third, we consider alternative criteria to eliminate outliers in our dependent variable. Columns (5)-(6) of Table 9 show the estimates of Equation (1) for two samples based on the following outlier criteria: excluding the top and bottom 10% of unit values for any product or replacing the observations with unit values above (below) the 75th (25th) percentile plus (minus) by 1.5 times the inter-quartile range by those cut-off values. The estimates show that the significant effect of declines in transport costs on quality upgrading is maintained.

Fourth, exporters as well as foreign-owned firms may produce higher-quality products thus exhibit higher unit values relative to domestic-owned firms, regardless of import competition. Columns (7)-(8) of Table 9 show that the estimate of  $\beta$  is robust to the addition of an indicator for the firm's exporter status or the firm's foreign ownership status. Measuring competition in the domestic market is inherently difficult. Column (9) of Table 9 shows that the effect of import competition is robust to using as measure of competition the sum of the market shares of the 5 firms with the largest market shares in each of the 4-digit industries to which a firm's products belong.<sup>37</sup> Moreover, within-country costs of transportation, among several other factors, may give firms in certain regions stronger market power. Hence, we show in column (10) of Table 9 the results from a specification where we add regional Herfindahl indexes and market shares. Our estimate of  $\beta$  remains qualitatively unchanged.

<sup>36.</sup> We refer the reader to Kugler and Verhoogen (2008) for further details on this two-stage procedure, in particular on the non-identification of some of the firm-year fixed effects.

<sup>37.</sup> In unreported regressions we also find robust effects of transport costs when we replace the firm's market share in each of its 4-digit industries by that in each of its 5-digit or 6-digit industries, or in each of its 7-digit products. Also we find the results to be maintained when the firm market share is defined relative to the sum of domestic output and imports in the 4-digit industry in each year. These results are available from the authors upon request.

# V. ALTERNATIVE INTERPRETATIONS

So far we presented our findings as evidence that increased import competition across industries stimulated product quality upgrading by Chilean firms. However, our findings could be given alternative interpretations to which we turn next.

# V.1. Input Quality

One alternative interpretation for our results is that they are picking up the access to better production inputs and their positive impact on product quality. An important benefit from increased trade openness for firms in developing countries is indeed the access to greater variety and better quality of inputs (e.g. Grossman and Helpman, 1991; Rivera-Batiz and Romer, 1991; Goldberg *et al.*, 2008; Kugler and Verhoogen, 2009). According to the quality complementarity hypothesis proposed by Kugler and Verhoogen (2008) input quality and productivity jointly determine the quality of firm output. This suggests in our case that if lower transport costs across Chilean industries led (as would be expected) to an increase in the quality of inputs available to Chilean firms, this would be an important alternative explanation for the resulting upgrades in output quality.<sup>38</sup> To address this possibility we consider the role of intermediate inputs' transport costs for Chilean firms' quality choices. Transport costs on intermediate inputs are computed for

each 4-digit industry j as the following weighted average: 
$$TCInputs_{t}^{j} = \sum_{m=1}^{M} \alpha_{j}^{m} * TC_{t}^{m} \text{ where } TC_{t}^{m} \text{ is}$$

the transport costs measure on final output (used in the regressions so far) and  $\alpha_j^m$  is the share of intermediate inputs from industry m in the total intermediates used for production by industry j taken from the 1996 Chilean input-output table.<sup>39</sup>

Column (1) of Table 10 presents the results from estimating a variant of Equation (1) where we include only transport costs on intermediate inputs. Transport costs on intermediate inputs have a negative and significant effect on product quality. It is indeed the case that there is a positive impact from increased access to foreign inputs on quality upgrading by Chilean firms.

<sup>38.</sup> Our specifications control for the share of intermediate inputs used by Chilean firms that are imported, but that is insufficient to ensure that the input channel is not what underlies the negative coefficient on transport costs. The reason is that higher quality imported inputs and higher quality domestic inputs are likely to be complementary. So, if input expenditures grow with input quality, the imported input share could be unchanged but the underlying domestic and imported inputs used by firms could be of better quality with consequences for output quality.

<sup>39.</sup> The correlation between transport costs on final output and transport costs on intermediate inputs is 0.54.

To explore whether this implies that our main finding is in fact explained through the effects of inputs, we show in column (2) the results from a specification where both transport costs on final output as well as on intermediate inputs are included. The results show very clearly that while input quality driven by reduced transport costs on inputs may be an alternative channel for the impact of transport costs on quality it is certainly not the only one. Rather, the effects of import competition measured by the coefficient on transport costs on output are still significant. Notice that the p-value for a test in the difference across coefficients indicates that such difference is not significant: i.e. better input quality may help explain the quality upgrading effects of lower transport costs but does not eliminate the importance that stronger product market competition through imports had for Chilean firms.<sup>40</sup>

Table 10: Effects of Transport Costs on Unit Values – Inputs

	Dependent Variable: Log of Firm-Product Unit Value		
	(1)	(2)	
Transport Costs for Intermediate Inputs	-4.514***	-3.888**	
	(1.413)	(1.615)	
Transport Costs t-1		-1.509**	
		(0.749)	
Firm Cost Controls	Yes	Yes	
Other Firm Controls	Yes	Yes	
Industry Controls	Yes	Yes	
Firm*Product Fixed Effects	Yes	Yes	
3-Digit Industry*Year Fixed Effects	Yes	Yes	
P-value for F-Test of Difference in			
Coefficients across Transport Costs		0.20	
Measures			
Number of Observations	41964	41929	
R-Squared	0.97	0.97	

Notes: Robust standard errors clustered by 4-digit industry and year in parentheses. \*\*, and \* indicate significance at 1%, 5%, and 10% confidence levels, respectively. The construction of the measure of transport costs for intermediate inputs is described in the text. The firm cost controls, other firm controls, and industry controls included in the regressions are similar to those in column (5) of Table 4.

# V.2. An Alternative Argument Based on Hummels and Skiba (2004) and Vogel (2008)

The effect of lower transport costs on product quality upgrading might also be rationalised by a strategic product market positioning argument based on a combination of the predictions derived by Hummels and Skiba (2004) and by Vogel (2008). Hummels and Skiba (2004) derive a theoretical relation between transport costs and the quality composition of trade whereby a per unit transport cost raises the relative demand for high quality exports. For our purposes, this prediction would suggest that reductions in transport costs led to a decline in the quality of Chilean imports. To be precise, what Hummels and Skiba (2004) show is that the quality of exports varies positively with transport costs per unit, not with ad-valorem costs such

<sup>40.</sup> Schor (2004) and Amiti and Konings (2007) also find the effects of both input tariffs and output tariffs to be significant in their analysis of TFP for firms in Brazil and Indonesia, respectively.

as those we use in most specification. However, despite this fact we consider the possibility that ad-valorem transport costs could also affect the quality of exports. Vogel (2008) proposes a model of endogenous product choice where firms differentiate themselves horizontally (in product attributes or geographic space) and vertically (in quality space) so as to fill out the product space. Combining the two ideas one could argue that Chilean firms increased the quality of their products to fill out the product space as the quality of imports declined with reductions in transport costs. The effect of transport costs on product quality would in this case be the outcome of spatial competition rather than reflecting import competition per se.

In Section III, we showed that the quality of Chilean imports did not change substantially during our sample period based on unit values of imports at the 4-digit ISIC and exporting country level. This would suggest that the argument made above is unlikely to hold. But to probe further into that argument, we present in Table 11 the results from a specification where time-varying unit values of imports at the 6-digit HS product and exporting country level are explained by either ad-valorem or per unit transport costs at the same level of disaggregation. This specification follows that estimated by Hummels and Skiba (2004) – i.e. it includes exporting country GDP per capita – but controls for product-exporting country and year fixed effects given its panel dimension. IV estimation is used where the instrument for transport costs is fuel prices interacted with transport costs in the first sample year.<sup>41</sup> The reason for using IV is that since more expensive goods have more onerous handling requirements, transport costs may increase with export unit values, rather than the reverse. The results show that the relationship between transport costs of either type and unit values of Chilean imports is insignificant. Hence, lower transport costs do not seem to have changed the unit values of Chilean imports in a manner consistent with the Hummels and Skiba (2004) mechanism.

Table 11: Effects of Transport Costs on Import Unit Values

	Dependent Variable: Log of Unit Value of Imports at 6-digit HS Level		
	IV	IV	
	(1)	(2)	
Transport Costs at 6-digit HS Level t	-0.088		
	(0.269)		
Transport Costs per Unit at 6-digit HS Level t		0.00007	
		(0.00007)	
GDP per Capita of Exporting Country	Yes	Yes	
Country*6-digit HS Product Fixed Effects	Yes	Yes	
Year Fixed Effects	Yes	Yes	
Number of Observations	82640	78271	

Notes: Robust standard errors in parentheses. The instrument in column (1) is fuel prices interacted with transport costs in the first sample year and the instrument in column (2) is fuel prices interacted with transport costs per unit in the first sample year. The regressions are based on a sample covering only the top 10 exporters for each 6-digit HS product and year. The p-value of the F-test of excluded instruments is 0.00 and the p-value of the under-identification test (Kleibergen-Paap LM statistic) is 0.00 for the IV regressions in columns (1)-(2).

© OECD 2010 39

\_

<sup>41.</sup> Fuel prices are described in Appendix 2. Transport costs at the level of the 6-digit HS product and exporting country in the first sample year are used.

### V.3. Market Access

In the analysis we assume implicitly that a reduction on transport costs is equivalent to a unilateral trade liberalisation that increases imports. However, reductions in the transport costs of imports into Chile may be correlated with reductions in the transport costs faced by Chilean exports. Hence, an alternative interpretation of the negative effect of transport costs on quality is that they reflect the effects of a symmetric increase in export market access and not those of product market competition. Verhoogen (2008) shows that such an increase led to quality upgrading by Mexican firms.

To address this concern, we estimate Equation (1) considering only the sub-sample of Chilean firms that do not export during our sample period. Column (1) of Table 12 presents the corresponding results that show a negative and significant impact of transport costs on product quality. The same is obtained considering an even more restricted sub-sample of non-exporting firms that are fully domestically-owned in column (2). Interestingly, for these two sub-samples the impact of transport costs is substantially more negative than that in column (5) of Table 4 for the full sample of firms. This means that an increase in import competition elicits the strongest quality upgrading response from the firms that are less exposed to international competition through other channels such as exports or multinational parent linkages. Such firms may already have been forced to undertake quality upgrading explaining why increased import competition provides a weaker incentive for further upgrading.

Table 12: Effects of Transport Costs on Unit Values – Market Access

	Dependent Variable: Log of Firm-Product Unit Value		
	Sample of Non- Exporting Firms	Sample of Domestic Non-Exporting Firms	
	(1)	(2)	(3)
Transport Costs t-1	-2.337*** (0.880)	-2.457*** (0.894)	
Transport Costs t-1 * Exported Products			0.420 (0.689)
Transport Costs t-1* Non-Exported Products			-2.480*** (0.779)
Firm Cost Controls	Yes	Yes	Yes
Other Firm Controls	Yes	Yes	Yes
Industry Controls	Yes	Yes	Yes
Firm*Product Fixed Effects	Yes	Yes	Yes
3-Digit Industry*Year Fixed Effects	Yes	Yes	Yes
P-value for F-Test of Difference in Coefficients across Groups			0.00
Number of Observations	32227	31333	41929
R-Squared	0.97	0.97	0.97

Notes: Robust standard errors clustered by 4-digit industry and year in parentheses. \*\*, and \* indicate significance at 1%, 5%, and 10% confidence levels, respectively. The firm cost controls, other firm controls, and industry controls included in the regressions are similar to those in column (5) of Table 4. In column (3) exported products are defined by a time-varying dummy that indicates whether a firm exports the product in a given year.

To address the issue of export market access further, we estimate a variant of Equation (3) defining group 1 to include products that are exported (at least partially) and group 2 to include

products that are sold exclusively in the domestic market. The results are shown in column (3) of Table 12. Interestingly, the estimates and F-test show that the impact of transport costs on quality is significantly more negative for domestically sold products. It is possible that once these domestically sold products achieve sufficiently high quality, firms are able to sell them in export markets also, which is indeed the finding for Mexican firms by Iacovone and Javorcik (2008). This result points to the importance of using data both for domestically sold as well as exported products to study the link between import competition and quality. Overall our findings suggest that we can exclude the possibility that the negative effect of transport costs on product quality is due to an export market access effect as the evidence points to benefits mainly for firms with no exporting experience and for products that are not exposed to export markets.

# VI. CONCLUSIONS AND POLICY RELEVANCE

So, does trade stimulate innovation? We investigate this question using a rich dataset of Chilean firms and products and a regression framework where increases in unit values proxy for product quality improvements and transport costs are the exogenous measure of import competition. Our results show that tougher import competition does have a positive, significant, and robust impact on incremental innovation reflected in product quality upgrading by Chilean firms. To the extent that these findings can be generalised to other emerging economies, they suggest that increased exposure to imports can be beneficial for innovation outcomes. Moreover, we find that the mechanism driving this outcome is that firms react to the import pressure by innovating so as to differentiate their products as a way to escape competition. Our findings, therefore, point to the importance of competition policy more generally. In addition, our results indicate that easier access to imported inputs also has beneficial effects on innovation, which points to the importance of learning from trade in stimulating innovation. However, our evidence also suggests that such benefits only arise if the right conditions hold: it requires firms to dispose of skilled personnel and will occur mostly in industries whose attributes offer opportunities for such innovativeness. This implies that for the dynamic benefits of trade to materialize the framework conditions do matter. A dimension that goes beyond what is studied in this paper is that, obviously, firms undertaking innovation may transform their production processes and this could have consequences for labor markets that may need to be addressed. Our evidence suggests, along the lines of Verhoogen (2008) that import competition may result in an increase in within-firm wage inequality if quality upgrading itself requires an increase in the demand for skilled labor. Two other aspects are left for future research. On the empirical side, studying the heterogeneity in firms' reactions to import competition is relevant. On the theoretical side, our findings suggest that the recent models of multi-product firms such as those of Bernard et al. (2006b), Eckel and Neary (2006), and Volker and Yeaple (2008) that examine changes in firms' product mix as a response to reductions in trade costs have yet to exploit other interesting margins of adjustment such as the possibility of quality upgrading.

# **ANNEXES**

**Appendix 1: Examples of 7-Digit Products for Selected 4-digit Industries** 

4-digit		7-digit	Product Description	Unit of	Average Annual
ISIC		ISIC		Measurement	Unit Value
					Changes
3117	Manufacture of bakery	3117101	Bread of any kind, size and quality (except sweet bread)	in tons	2.89%
	products	3117201	Cookies, with and without sugar and filled	in tons	4.28%
		3117301	Noodles, pasta including macaroni	in tons	0.22%
		3117402	Mixed dough (for different types of cakes)	in tons	-11.42%
3311	Sawmills, planing and other	3311307	Finished parquet excluding plastic parquet	in square meters	1.28%
	wood mills	3311302	Wooden boards for prefabricated houses	in square meters	-13.16%
		3311306	Wooden doors with or without glass	in units	0.10%
		3311124	Sawing wood	in cubic meters	5.94%
3320	Manufacture of furniture and	3320908	Sofas and armchairs of the type used in ceremonies	in units	31.79%
	fixtures, except primarily of	3320910	Wooden tables for computers and typewriters	in units	10.08%
	metal	3320906	Wooden household furniture	in units	26.47%
		3320913	Office furniture	in units	-5.45%
3483	Manufacture of motor vehicles	3843201	Fabricated motor vehicles	in units	0.41%
		3843409	Wheels and related parts and vehicle accessories	in units	5.78%
		3843421	Heating appliances for motor vehicles	in units	-2.68%
		3843422	Metallic frames for trucks, special frames	in units	19.27%
3559	Manufacture of rubber	3559324	Gloves of caoutchouc	one pair	13.63%
	products n.e.c.	3559327	Sports shoes	one pair	5.51%
		3559320	Caoutchouc sheating for mining	in tons	17.00%
		3559332	Articles made of caoutchouc for vehicles	in tons	26.81%
3829	Machinery and equipment	3829056	Cablecars	in units	20.74%
	except electrical n.e.c.	3829032	Gas regulators	in units	-4.56%
		3829060	Moving staircases	in units	26.01%
		3829002	Pumps for liquids for manual use	in units	13.71%

Notes: For each 7-digit product and year, we compute the average logarithmic unit value by pooling across all firms that manufacture that product. Then across any two consecutive years we compute the difference in average log unit values to obtain the annual change in unit values. The statistic in the table shows the simple average of those annual changes.

### **Appendix 2: Data Issues**

# Appendix 2.1: Firm and Products Data

We combine a products dataset at the 7-digit level for the period 1997-2003 and the annual manufacturing census of Chilean firms with more than 10 employees (ENIA) for the same period. In the products dataset, products are identified by a classification based on ISIC Rev. 2 and Rev. 3. More detail on the products data is provided in Navarro (2008). We obtain products at the 7-digit level building up from what Navarro (2008) refers to as 'ENIA products'.

Specifically, for each firm reporting more than one entry for a 7-digit product in a given year (Z entries) we sum the information on sales values and product quantities of those Z entries for that firm as long as all the Z entries' quantities are reported in the same unit. The sum provides us with a single entry for that 7-digit product for that firm in that year. If the entries' quantities are reported in multiple units, we drop those products from the analysis. Note that these deletions occur in a very small number of cases. Also note that if aggregated to the 4-digit level, our 7-digit products correspond exactly to the United Nations product classification.

For our analysis, we use information on sales values and product quantities sold for each 7-digit product, firm, and year. We exclude from the final sample (i) firms that do not report the measurement unit for their products' quantities and (ii) firms that report their products' quantities in a different unit than the unit in which the majority of firms report. We also exclude from the sample the observations with unit values above (below) the 75th (25th) percentile plus (minus) by 1.5 times the inter-quartile range of the unit values' distribution for any 7-digit product. After applying these data cleaning procedures our final sample includes 51 349 firm-year-product observations.

We test the goodness of our products data by identifying firms with irregular product 'drops' (i.e. products that disappear from production and then reappear again) and firms with product 'jumps' (i.e. products that are produced only once in the intermediate years of firm presence in the sample). These tests, which follow Bernard *et al.* (2008), are satisfactory in that product 'drops' and product 'jumps' are relatively infrequent. We also perform another test which compares the standard deviations of 'purged' unit values for 4-digit industries with the same standard deviations obtained for a Colombian products dataset by Kugler and Verhoogen (2008). 'Purged unit values' are the residuals from regressions of log unit values on product fixed effects or from regressions of log unit values on product-year fixed effects. Our standard deviations are somewhat larger than theirs but are sufficiently within bounds to be explained by the fact that we consider a different country with a distinct profile of manufacturing production.

We use variables from the ENIA census to compute the proxies for costs of production included in our regressions. Firm average wages are obtained as the ratio of total wages paid to the firm's employees. Firm skill share is defined as the ratio of the number of skilled workers (a sum of managers, administrative personnel and qualified production workers) to the total number of workers employed by the firm. Firm electricity unit prices are computed as the log of the ratio of electricity expenditure to the quantity of electricity purchased. To eliminate outliers in each of these variables, we follow a 'winsorising' procedure whereby we replace the top and bottom 5th percentile of observations in each year by the value of the cut-off observations at the 5th and 95th percentile in that year, respectively. Firm share of imported materials is computed as the ratio of the expenditure in imported materials and primary inputs to the overall expenditure in materials and primary inputs. The three size dummies are defined based on total employment: small firms have less than 50 employees, medium firms have 50 to 200 employees, and large firms have more than 200 employees.

Regarding the industry control variables included in the regressions, since total employment of a firm is not allocated across the production of each of its products, the share of total employment accounted for by foreign-owned firms is computed for the firm's main 4-digit industry, which is for multi-product firms the industry to which the major product belongs. The

major product accounts for the largest share (which could be less than 50%) of the firm's total sales.

### Appendix 2.2: Transport Costs Data

We use a transport costs dataset from the ALADI secretariat for the period 1997-2003 that includes the freight value (excluding insurance costs) and the free on board customs value (fob) of Chilean imports for each 8 digit HS code, exporting country, and year. For each 8-digit HS code, exporting country, and year we compute a freight rate as the ratio of the freight costs to the fob imports. Our measure of transport costs is given by a weighted average of the freight rate aggregated from the level of the 8-digit HS code, exporting country, and year, to the level of the 4-digit ISIC and year using as weights Chile's fob imports from each country and year. To convert import flows between 8-digit HS codes and 4-digit ISIC codes we use a correspondence obtained from http://www.macalester.edu/research/economics

/PAGE/ HAVEMAN/Trade.Resources/TradeConcordances.html. Our dataset includes all Chilean imports originating in 169 countries. Taking the overall value of imports for the entire period 1997-2003, the top 10 exporters to Chile are the United States, Brazil, Argentina, China, Germany, Japan, France, Mexico, South Korea, and Italy.

### Appendix 2.3: Import Penetration Data and Instruments

The import penetration ratio of a 4-digit ISIC industry is computed as the ratio between total imports of the industry and the sum of total imports and total domestic sales of the industry. Imports at the 4-digit ISIC level are taken from the COMTRADE database and total domestic sales are the sum of total sales across firms from the ENIA dataset.

The two instruments for transport costs used in the regressions of import penetration on transport costs whose results are reported in Table 3 are defined making use of fuel prices obtained – following Backus and Crucini (2000) – as the simple average of spot crude prices in three major markets (Brent, Dubai, and West Texas) taken from IEA (2008). The instrument used in column (2) is the interaction between fuel prices and the 8-digit HS product's freight rate in the first sample year which is aggregated up to the 4-digit and year level as a weighted average using the aforementioned concordance and weights that are the share of each 8-digit HS product and exporting country in Chile's total imports in the corresponding 4-digit ISIC industry in the first sample year. The instrument used in column (3) is the interaction between fuel prices and a distance measure which is aggregated up to the 4-digit and year level as a weighted average using the aforementioned concordance and weights that are the share of each 8-digit HS product and exporting country in Chile's total imports in the corresponding 4-digit ISIC industry in the first sample year. The distance measure is taken from **CEPII** (http://www.cepii.fr/anglaisgraph/bdd/distances.htm) and consists of geodesic distances calculated following the great circle formula, which uses latitudes and longitudes of the most important cities/agglomerations (in terms of population). By first sample year we mean the first year when an 8-digit HS product and exporting country pair appears in the ALADI dataset.

### Appendix 2.4: Industry-Specific Exchange Rates

Industry-specific exchange rates are computed following Goldberg (2004) as the weighted average of the bilateral nominal exchange rates between Chile and each of its trading partners where the weights are given by the share of each partner country in total Chilean exports for a given 4-digit ISIC industry prior to our sample period in 1996. Bilateral nominal exchange rates are computed based on International Financial Statistics data from the IMF and export shares are based on COMTRADE data at the 4-digit ISIC level.

# Appendix 3: Methodology and Data Issues for Price-Cost Margins

The difference between the primal Solow residual and the corresponding dual Solow residual derived from a cost function results in the equation below which follows Konings *et al.* (2005):

$$\left(\frac{\Delta Y_{it}}{Y_{it}} + \frac{\Delta P_{Yit}}{P_{Yit}}\right) - \alpha_{Lit} \left(\frac{\Delta L_{it}}{L_{it}} + \frac{\Delta P_{Lit}}{P_{Lit}}\right) - \alpha_{Mit} \left(\frac{\Delta M_{it}}{M_{it}} + \frac{\Delta P_{Mit}}{P_{Mit}}\right) - (1 - \alpha_{Lit} - \alpha_{Mit}) \left(\frac{\Delta K_{it}}{K_{it}} + \frac{\Delta P_{Kit}}{P_{Kit}}\right) \\
= \beta_{it} \left[\left(\frac{\Delta Y_{it}}{Y_{it}} + \frac{\Delta P_{Yit}}{P_{Yit}}\right) - \left(\frac{\Delta K_{it}}{K_{it}} + \frac{\Delta P_{Kit}}{P_{Kit}}\right)\right] \tag{A1}$$

where  $\beta_{it}$  is the price-cost margin for firm i in year t,  $\left(\Delta Y_{it}/Y_{it} + \Delta P_{Y_{it}}/P_{Y_{it}}\right)$  is nominal sales growth,  $\left(\Delta L_{it}/L_{it} + \Delta P_{Lit}/P_{Lit}\right)$  is wage bill growth,  $\left(\Delta M_{it}/M_{it} + \Delta P_{Mit}/P_{Mit}\right)$  is intermediate costs growth,  $\left(\Delta K_{it}/K_{it} + \Delta P_{Kit}/P_{Kit}\right)$  is capital stock growth, and  $\alpha_{Lit}$ ,  $\alpha_{Mit}$  are labor and intermediates shares in total nominal sales. Equation (A1) assumes constant returns to scale:  $(1-\alpha_{Lit}-\alpha_{Mit})$  is the cost share of capital. To reach Equation (2) in the text we designate the left hand side of Equation (A1) by  $\Delta Z_{it}$ , and the right hand side parentheses term by  $\Delta X_{it}$ , we interact  $\Delta X_{it}$  separately with the transport costs measure and with the Herfindahl index, we include in Equation (A1) the transport costs measure and the Herfindahl index levels as well as year fixed effects and we add an i.i.d. stochastic residual  $\eta_{it}$ . Equation (2) in the text is estimated for the sample of firms in the ENIA dataset during the 1997-2003 period. For firms with discontinuous data we include only the observations across consecutive years for which yearly growth rates of variables can be computed. The sample differs from that used for the unit values regressions since the observations are dropped based on the following criteria: i) we exclude from the sample firms with missing sales, wage bill, intermediate costs, or capital variables; ii) we impute sales, wage bill, intermediate costs, or capital to correct for non-reporting by a firm in a single year (which occurs in fewer than 30 firm-year observations); iii) we exclude from the sample firms whose sales growth, wage bill growth, or capital growth is larger than (smaller than) 400%; iv) we exclude from the sample firms whose sales (wage bill) growth ranges between 100% and 300% (-300% and -100%) but is not accompanied by corresponding high (low) growth rates of intermediate costs (total employment). After applying these data cleaning procedures our final sample includes 31 318 firm-year observations.

To compute  $\Delta Z_{ii}$  and  $\Delta X_{ii}$ , we use firm-level information on nominal sales and on total wage bill and compute their corresponding logarithmic growth rates. Nominal intermediate costs are obtained as the sum of materials costs and electricity costs and the corresponding

logarithmic growth rate is calculated. Capital stocks are computed using the perpetual inventory method (PIM) as described in Fernandes and Paunov (2008) and the corresponding logarithmic growth rate is computed. We define the rental price of capital to be equal to the product of the aforementioned investment goods price deflator and the sum of the real interest rate and a depreciation rate as in Konings *et al.* (2005). Similarly, data on the lending interest rate and the consumer price index taken from the IMF financial statistics is used to compute the real interest rate. The depreciation rate used is the simple average of the rates used by Fernandes and Paunov (2008) for three types of capital goods: 3% for buildings, 7% for machinery and equipment, and 11.9% for transport equipment. Using an alternative depreciation rate equal to 10% provides almost similar results. The share of labor (*intermediates*) in sales is given by the ratio of the wage bill (*intermediate costs*) to total nominal sales.

### **REFERENCES**

- AGHION, P. and P. HOWITT (1998), Endogenous Growth Theory, MIT Press.
- AGHION, P., N. BLOOM, R. BLUNDELL, R. GRIFFITH and P. HOWITT (2005), "Competition and Innovation: An Inverted-U Relationship", *Quarterly Journal of Economics* 120(2): 701-728.
- AGHION, P., R. BLUNDELL, R. GRIFFITH, P. HOWITT and S. PRANTL (2006), "The Effects of Entry on Incumbent Innovation and Productivity", NBER Working Paper No. 12027.
- AHN, S., (2002), "Competition, Innovation and Productivity Growth: A Review of Theory and Evidence", OECD, Economics Department Working Paper No. 317.
- AKERLOF, G. (1970), "The Market for 'Lemons': Qualitative Uncertainty and the Market Mechanism", *Quarterly Journal of Economics* 84(3): 488-500.
- AMITI, M. and J. KONINGS (2007), "Trade Liberalization, Intermediate Inputs, and Productivity: Evidence from Indonesia" *American Economic Review* 97(5): 1611-1638.
- ANDERSON, J. E. and E. van WINCOOP (2004), "Trade Costs", Journal of Economic Literature 42(3): 691-751.
- ATKESON, A. and A. BURSTEIN (2007), "Innovation, Firm Dynamics, and International Trade", NBER Working Paper No. 13326.
- AW, B., X. CHEN and M. ROBERTS (2001), "Firm-Level Evidence on Productivity Differentials and Turnover in Taiwanese Manufacturing", *Journal of Development Economics* 66(1): 51-86.
- BACKUS and CRUCINI (2000), "Oil Prices and the Terms of Trade", Journal of International Economics 50(1) 185-213.
- BALDWIN, J. and W. Gu (2004), "Trade Liberalization: Export-market Participation, Productivity Growth, and Innovation", Oxford Review of Economic Policy 20(3): 372-392.
- BASTOS, P. and J. SILVA (2009), "Identifying Vertically Differentiated Products", *Economic Letters* forthcoming.
- BENAVENTE, J. M., CRESPI, G. and A. MAFFIOLI (2007), "Public Support to Firm-Level Innovation: An Evaluation of the FONTEC Program", *OVE Working Papers* 0507, Inter-American Development Bank, Office of Evaluation and Oversight (OVE).
- BERNARD, A., JENSEN, J. and P. SCHOTT (2006a), "Trade Costs, Firms, and Productivity", *Journal of Monetary Economics* 53(5): 917-937.
- BERNARD, A., S.REDDING and P. SCHOTT (2006b), "Multi-Product Firms and Trade Liberalization", NBER Working Paper No. 12782.
- BERNARD, A., S. REDDING and P. SCHOTT (2009), "Multi-Product Firms and Product Switching", *American Economic Review* forthcoming.
- BERTSCHEK, I. (1995). "Product and Process Innovation as a Response to Increasing Imports and Foreign Direct Investment", *Journal of Industrial Economics* 43(4): 341-357.
- BLOOM, N., M. DRAKA and J. VAN REENEN (2009a), "Trade Induced Technical Change? The Impact of Chinese Imports on Innovation, Diffusion and Productivity", LSE mimeo.
- BLOOM, N., P. ROMER and J. VAn REENEN (2009b), "A Trapped Factor Model of Innovation", LSE mimeo.
- BOULHOL, H. (2009), "Pro-Competitive Effect of Trade and Non-Decreasing Price-Cost Margins", Oxford Bulletin of Economics and Statistics, forthcoming.

- BRODA, C., J. GREENFIELD and D. WEINSTEIN (2006), "From Groundnuts to Globalization: A Structural Estimate of Trade and Growth", NBER Working Paper No. 12512.
- BROOKS, E. (2006), "Why Don't Firms Export More? Product Quality and Colombian Plants", *Journal of Development Economics* 80(1): 160-178.
- BUSTOS, P. (2009), "Trade Liberalization, Exports and Technology Upgrading: Evidence on the Impact of MERCOSUR on Argentinean Firms", *American Economic Review* forthcoming.
- CHUMACERO, R., R. FUENTES and K. SCHMIDT-HEBBEL (2004), "Chile's Free Trade Agreements: How Big is the Deal?" Central Bank of Chile Working Paper 264.
- COHEN, W. and S. KLEPPER, (1996), "Firm Size and the Nature of Innovation within Industries: The Case of Process and Product R&D", *Review of Economics and Statistics* 78(2): 232-243.
- COHEN, W. and D. LEVINTHAL (1989), "Innovation and Learning: the Two Faces of R&D", Economic Journal 99(397): 569-596.
- ECKEL, C. and P. NEARY (2006), "Multi-Product Firms and Flexible Manufacturing in the Global Economy", CEPR Discussion Paper 5941.
- EDERINGTON, J. and P. MACCALMAN (2008), "Endogenous Firm Heterogeneity and the Dynamics of Trade Liberalization", *Journal of International Economics* 74(2): 422-440.
- FERNANDES, A. (2007), "Trade Policy, Trade Volumes, and Plant-Level Productivity in Colombian Manufacturing Industries", *Journal of International Economics* 71(1): 52-71.
- FERNANDES, A. and C. PAUNOV (2008), "Foreign Direct Investment in Services and Manufacturing Productivity Growth: Evidence for Chile", Policy Research Working Paper 4730.
- FLUET, C and P. GARELLA (2002), "Advertising and Prices as Signals of Quality in a Regime of Price Rivalry", *International Journal of Industrial Organization* 20(7): 907-930.
- FONTAGNÉ, L. and M. FREUDENBERG (1997), "Intra-Industry Trade: Methodological Issues Reconsidered", CEPII Document de Travail n° 97-01.
- GOH, A.-T. (2000), "Opportunity Cost, Trade Policies and the Efficiency of Firms", *Journal of Development Economics* 62(2): 363-383.
- GOLDBERG, P., A. KHANDELWAL, N. PAVCNIK and P. TOPALOVA (2008), "Imported Intermediate Inputs and Domestic Product Growth: Evidence from India", NBER Working Paper No. 14416.
- GOLDBERG, P., KHANDELWAL, A., PAVCNIK, N. and P. TOPALOVA (2009), "Multi-Product Firms and Product Turnover in the Developing World: Evidence From India", *Review of Economics and Statistics* forthcoming.
- GRIFFITH, R., R. HARRISON and J. VAN REENEN (2006), "How Special Is the Special Relationship? Using the Impact of U.S. R&D Spillovers on UK Firms as a Test of Technology Sourcing", *American Economic Review* 96(5): 1859-1875.
- GROSSMAN, G. and E. HELPMAN (1991), Innovation and Growth in the Global Economy, MIT Press, Cambridge, Massachussets.
- HALL, R. (1988), "The Relation Between Price and Marginal Cost in U.S. Industry", NBER Working Paper No. 1785.
- HARRISON, A. (1994), "Productivity, Imperfect Competition and Trade Reform: Theory and Evidence", *Journal of International Economics* 36(1-2): 53-73.
- HUMMELS, D. (2001), "Toward a Geography of Trade Costs", Purdue University mimeo.
- HUMMELS, D. (2007), "Transportation Costs and International Trade in the Second Era of Globalization", *Journal of Economic Perspectives* 21(3): 131-154.
- HUMMELS, D. and P. KLENOW (2005), "The Variety and Quality of a Nation's Exports", *American Economic Review* 95(3): 704-723.
- HUMMELS, D., V. LUGOVSKYY and A. SKIBA (2009), "The Trade Reducing Effects of Market Power in International Shipping", *Journal of Development Economics* 89(1): 84-97.

- HUMMELS, D. and A. SKIBA (2004), "Shipping the Good Apples Out? An Empirical Confirmation of the Alchian-Allen Conjecture", *Journal of Political Economy* 112(6): 1384-1402.
- IACOVONE, L. and B. JAVORCIK (2008), "Shipping Good Tequila Out: Investment, Domestic Unit Values and Entry of Multi-Product Plants into Export Markets", University of Oxford mimeo.
- IEA (2008), Annual Statistical Supplement to the Oil Market Report.
- JOHNSON, R. (2009), "Trade and Prices with Heterogeneous Firms", Dartmouth College mimeo.
- KHANDELWAL, A. (2009), "The Long and Short (of) Quality Ladders", Review of Economic Studies forthcoming.
- KIYOTA, K. (2008), "Are U.S. Exports Different from China's Exports? Evidence from Japan's Imports", Gerald R. Ford School of Public Policy University of Michigan, Discussion Paper No. 576.
- KONINGS, J., P. VAN CAYSEELE and F. WARZYNSKI (2001), "The Dynamics of Industrial Mark-Ups in Two Small Open Economies: Does National Competition Policy Matter?", *International Journal of Industrial Organization* 19(5): 841-859.
- KONINGS, J., P. VAN CAYSEELE and F. WARZYNSKI (2005), "The Effects of Privatization and Competitive Pressure on Firms' Price-Cost Margins: Micro Evidence from Emerging Economies", Review of Economics and Statistics 87(1): 124-134.
- KUGLER, M. and E. VERHOGEN (2008), "Product Quality at the Plant Level: Plant Size, Exports and Product Prices in Colombia", NBER Working Paper No. 14418.
- KUGLER, M. and E. VERHOOGEN (2009), "Plants and Imported Inputs: New Facts and an Interpretation", American Economic Review Papers and Proceedings 99(2): 501-507.
- Lelarge, C. and B. Nefussi (2008), "Exposure to Low-Wage Competition, Activity Changes and Quality Upgrading: An Empirical Assessment", mimeo.
- LEVINSOHN, J. (1993), "Testing the Imports-as-Market-Discipline Hypothesis", *Journal of International Economics* 35(1-2): 1-22.
- LIMÃO, N. and A. VENABLES (2001), "Infrastructure, Geographical Disadvantage, Transport Costs, and Trade", World Bank Economic Review 15(3): 451-479.
- MANOVA, K. and Z. ZHANG (2009), "Export Prices and Heterogeneous Firm Models", Stanford University mimeo
- MELITZ, M. and G. OTTAVIANO (2008), "Market Size, Trade, and Productivity", Review of Economic Studies 75(1): 295-316.
- MOREIRA, M. and J. BLYDE (2006), "Chile's Integration Strategy: Is There Room for Improvement?" INTAL-ITD Working Paper 21.
- NAVARRO, L. (2008), "Plant Level Evidence on Product Mix Changes in Chilean Manufacturing", Queen Mary, University of London mimeo.
- NICKELL, S. (1996), "Competition and Corporate Performance", Journal of Political Economy 104(4): 724-746.
- NOCKE, V. and S. YEAPLE (2008), "Globalization and the Size Distribution of Multiproduct Firms", CEPR Discussion Paper 6948.
- OECD (1997), Oslo Manual: Proposed Guidelines for Collecting and Interpreting Technological Innovation Data, OECD, Paris.
- PAVCNIK, N. (2002), "Trade Liberalization, Exit, and Productivity Improvements: Evidence from Chilean Plants", *Review of Economic Studies* 69(1): 245-276.
- PIETROBELLI, C. and R. RABELLOTTI (2006), *Upgrading to Compete*. Inter-American Development Bank, David Rockefeller Center for Latin American Studies, Harvard University.
- PUGA, D. and D. TREFLER (2010), "Wake Up and Smell the Ginseng: International Trade and the Rise of Incremental Innovation in Low-Wage Countries", *Journal of Development Economics* 91(1): 64-76.
- RAUCH, J. (1999). "Network versus Markets in International Trade", *Journal of International Economics* 48(1): 7-35.

- RIVERA-BATIZ, L. and P. ROMER (1991). "Economic Integration and Endogenous Growth", *Quarterly Journal of Economics* 106(2): 531-555.
- ROBERTS, M. and J. Tybout (1996). Industrial Evolution in Developing Countries: Micro Patterns of Turnover, Productivity, and Market Structure, World Bank, Washington, DC.
- RODRIK, D. (1992), "Closing the Productivity Gap: Does Trade Liberalization Really Help", in G. Helleiner (ed.) *Trade Policy, Industrialization and Development*, Clarendon Press, Oxford.
- ROEGER, W. (1995), "Can Imperfect Competition Explain the Difference Between Primal and Dual Productivity Measures? Estimates from U.S. Manufacturing", *Journal of Political Economy* 103(2): 316-330.
- SCHMOCH, U, F. LAVILLE and R. FRIETSCH (2003), "Linking Technology Areas to Industrial Sectors", Final Report to the European Commission, DG Research.
- SCHOR, A. (2004), "Heterogeneous Productivity Response to Tariff Reduction. Evidence from Brazilian Manufacturing Firms", *Journal of Development Economics* 75(2): 373-396.
- Schott, P. (2008), "The Relative Sophistication of Chinese Exports", Economic Policy 23(53): 5-49.
- SCHUMPETER, J. (1942), Capitalism, Socialism, and Democracy, Harper, New York.
- TESHIMA, K. (2009), "Import Competition and Innovation at the Plant Level: Evidence from Mexico", Columbia University mimeo.
- THOMAS, L., SHANE, S. and K. WEIGELT (1998), "An Empirical Examination of Advertising as a Signal of Product Quality", *Journal of Economic Behavior and Organization* 37(4): 415-430.
- THOENIG, M. and T. VERDIER (2003), "A Theory of Defensive Skill-Based Innovation and Globalization", *American Economic Review* 93(3): 709-728.
- Tybout, J. (2000), "Manufacturing Firms in Developing Countries: How Well Do They Do, and Why?", Journal of Economic Literature 38(1): 11-44.
- VERHOOGEN, E. (2008), "Trade, Quality Upgrading, and Wage Inequality in the Mexican Manufacturing Sector", *Quarterly Journal of Economics* 123(2): 489-530.
- VOGEL, J. (2008), "Spatial Competition with Heterogeneous Firms", Journal of Political Economy 116(3): 423-466.

# OTHER TITLES IN THE SERIES/ AUTRES TITRES DANS LA SÉRIE

The former series known as "Technical Papers" and "Webdocs" merged in November 2003 into "Development Centre Working Papers". In the new series, former Webdocs 1-17 follow former Technical Papers 1-212 as Working Papers 213-229.

All these documents may be downloaded from: http://www.oecd.org/dev/wp or obtained via e-mail (dev.contact@oecd.org).

Working Paper No.1, Macroeconomic Adjustment and Income Distribution: A Macro-Micro Simulation Model, by François Bourguignon, William H. Branson and Jaime de Melo, March 1989.

Working Paper No. 2, International Interactions in Food and Agricultural Policies: The Effect of Alternative Policies, by Joachim Zietz and Alberto Valdés, April, 1989.

Working Paper No. 3, The Impact of Budget Retrenchment on Income Distribution in Indonesia: A Social Accounting Matrix Application, by Steven Keuning and Erik Thorbecke, June 1989.

Working Paper No. 3a, Statistical Annex: The Impact of Budget Retrenchment, June 1989.

Document de travail No. 4, Le Rééquilibrage entre le secteur public et le secteur privé : le cas du Mexique, par C.-A. Michalet, juin 1989.

Working Paper No. 5, Rebalancing the Public and Private Sectors: The Case of Malaysia, by R. Leeds, July 1989.

Working Paper No. 6, Efficiency, Welfare Effects, and Political Feasibility of Alternative Antipoverty and Adjustment Programs, by Alain de Janvry and Elisabeth Sadoulet, December 1989.

Document de travail No. 7, Ajustement et distribution des revenus : application d'un modèle macro-micro au Maroc, par Christian Morrisson, avec la collaboration de Sylvie Lambert et Akiko Suwa, décembre 1989.

Working Paper No. 8, Emerging Maize Biotechnologies and their Potential Impact, by W. Burt Sundquist, December 1989.

Document de travail No. 9, Analyse des variables socio-culturelles et de l'ajustement en Côte d'Ivoire, par W. Weekes-Vagliani, janvier 1990. Working Paper No. 10, A Financial Computable General Equilibrium Model for the Analysis of Ecuador's Stabilization Programs, by André Fargeix and Elisabeth Sadoulet, February 1990.

Working Paper No. 11, Macroeconomic Aspects, Foreign Flows and Domestic Savings Performance in Developing Countries: A "State of The Art" Report, by Anand Chandavarkar, February 1990.

Working Paper No. 12, Tax Revenue Implications of the Real Exchange Rate: Econometric Evidence from Korea and Mexico, by Viriginia Fierro and Helmut Reisen, February 1990.

Working Paper No. 13, Agricultural Growth and Economic Development: The Case of Pakistan, by Naved Hamid and Wouter Tims, April 1990.

Working Paper No. 14, Rebalancing the Public and Private Sectors in Developing Countries: The Case of Ghana, by H. Akuoko-Frimpong, June 1990.

Working Paper No. 15, Agriculture and the Economic Cycle: An Economic and Econometric Analysis with Special Reference to Brazil, by Florence Contré and Ian Goldin, June 1990.

Working Paper No. 16, Comparative Advantage: Theory and Application to Developing Country Agriculture, by Ian Goldin, June 1990.

Working Paper No. 17, Biotechnology and Developing Country Agriculture: Maize in Brazil, by Bernardo Sorj and John Wilkinson, June 1990.

Working Paper No. 18, Economic Policies and Sectoral Growth: Argentina 1913-1984, by Yair Mundlak, Domingo Cavallo, Roberto Domenech, June 1990.

Working Paper No. 19, Biotechnology and Developing Country Agriculture: Maize In Mexico, by Jaime A. Matus Gardea, Arturo Puente Gonzalez and Cristina Lopez Peralta, June 1990.

Working Paper No. 20, Biotechnology and Developing Country Agriculture: Maize in Thailand, by Suthad Setboonsarng, July 1990.

Working Paper No. 21, International Comparisons of Efficiency in Agricultural Production, by Guillermo Flichmann, July 1990.

Working Paper No. 22, *Unemployment in Developing Countries: New Light on an Old Problem*, by David Turnham and Denizhan Eröcal, Iuly 1990.

Working Paper No. 23, Optimal Currency Composition of Foreign Debt: the Case of Five Developing Countries, by Pier Giorgio Gawronski, August 1990.

Working Paper No. 24, From Globalization to Regionalization: the Mexican Case, by Wilson Peres Núñez, August 1990.

Working Paper No. 25, Electronics and Development in Venezuela: A User-Oriented Strategy and its Policy Implications, by Carlota Perez, October 1990.

Working Paper No. 26, The Legal Protection of Software: Implications for Latecomer Strategies in Newly Industrialising Economies (NIEs) and Middle-Income Economies (MIEs), by Carlos Maria Correa, October 1990.

Working Paper No. 27, Specialization, Technical Change and Competitiveness in the Brazilian Electronics Industry, by Claudio R. Frischtak, October 1990.

Working Paper No. 28, Internationalization Strategies of Japanese Electronics Companies: Implications for Asian Newly Industrializing Economies (NIEs), by Bundo Yamada, October 1990.

Working Paper No. 29, The Status and an Evaluation of the Electronics Industry in Taiwan, by Gee San, October 1990.

Working Paper No. 30, The Indian Electronics Industry: Current Status, Perspectives and Policy Options, by Ghayur Alam, October 1990.

Working Paper No. 31, Comparative Advantage in Agriculture in Ghana, by James Pickett and E. Shaeeldin, October 1990.

Working Paper No. 32, Debt Overhang, Liquidity Constraints and Adjustment Incentives, by Bert Hofman and Helmut Reisen, October 1990.

Working Paper No. 34, Biotechnology and Developing Country Agriculture: Maize in Indonesia, by Hidjat Nataatmadja et al., January 1991.

Working Paper No. 35, Changing Comparative Advantage in Thai Agriculture, by Ammar Siamwalla, Suthad Setboonsarng and Prasong Werakarnjanapongs, March 1991.

Working Paper No. 36, Capital Flows and the External Financing of Turkey's Imports, by Ziya Önis and Süleyman Özmucur, July 1991.

Working Paper No. 37, The External Financing of Indonesia's Imports, by Glenn P. Jenkins and Henry B.F. Lim, July 1991.

Working Paper No. 38, Long-term Capital Reflow under Macroeconomic Stabilization in Latin America, by Beatriz Armendariz de Aghion, July 1991.

Working Paper No. 39, Buybacks of LDC Debt and the Scope for Forgiveness, by Beatriz Armendariz de Aghion, July 1991.

Working Paper No. 40, Measuring and Modelling Non-Tariff Distortions with Special Reference to Trade in Agricultural Commodities, by Peter J. Lloyd, July 1991.

Working Paper No. 41, The Changing Nature of IMF Conditionality, by Jacques J. Polak, August 1991.

Working Paper No. 42, Time-Varying Estimates on the Openness of the Capital Account in Korea and Taiwan, by Helmut Reisen and Hélène Yèches, August 1991.

Working Paper No. 43, Toward a Concept of Development Agreements, by F. Gerard Adams, August 1991.

Document de travail No. 44, Le Partage du fardeau entre les créanciers de pays débiteurs défaillants, par Jean-Claude Berthélemy et Ann Vourc'h, septembre 1991.

Working Paper No. 45, The External Financing of Thailand's Imports, by Supote Chunanunthathum, October 1991.

Working Paper No. 46, The External Financing of Brazilian Imports, by Enrico Colombatto, with Elisa Luciano, Luca Gargiulo, Pietro Garibaldi and Giuseppe Russo, October 1991.

Working Paper No. 47, Scenarios for the World Trading System and their Implications for Developing Countries, by Robert Z. Lawrence, November 1991.

Working Paper No. 48, Trade Policies in a Global Context: Technical Specifications of the Rural/Urban-North/South (RUNS) Applied General Equilibrium Model, by Jean-Marc Burniaux and Dominique van der Mensbrugghe, November 1991.

Working Paper No. 49, Macro-Micro Linkages: Structural Adjustment and Fertilizer Policy in Sub-Saharan Africa, by Jean-Marc Fontaine with the collaboration of Alice Sindzingre, December 1991.

Working Paper No. 50, Aggregation by Industry in General Equilibrium Models with International Trade, by Peter J. Lloyd, December 1991.

Working Paper No. 51, Policy and Entrepreneurial Responses to the Montreal Protocol: Some Evidence from the Dynamic Asian Economies, by David C. O'Connor, December 1991.

Working Paper No. 52, On the Pricing of LDC Debt: an Analysis Based on Historical Evidence from Latin America, by Beatriz Armendariz de Aghion, February 1992.

Working Paper No. 53, Economic Regionalisation and Intra-Industry Trade: Pacific-Asian Perspectives, by Kiichiro Fukasaku, February 1992.

Working Paper No. 54, Debt Conversions in Yugoslavia, by Mojmir Mrak, February 1992.

Working Paper No. 55, Evaluation of Nigeria's Debt-Relief Experience (1985-1990), by N.E. Ogbe, March 1992.

Document de travail No. 56, L'Expérience de l'allégement de la dette du Mali, par Jean-Claude Berthélemy, février 1992.

Working Paper No. 57, Conflict or Indifference: US Multinationals in a World of Regional Trading Blocs, by Louis T. Wells, Jr., March 1992.

Working Paper No. 58, Japan's Rapidly Emerging Strategy Toward Asia, by Edward J. Lincoln, April 1992.

Working Paper No. 59, The Political Economy of Stabilization Programmes in Developing Countries, by Bruno S. Frey and Reiner Eichenberger, April 1992.

Working Paper No. 60, Some Implications of Europe 1992 for Developing Countries, by Sheila Page, April 1992.

Working Paper No. 61, Taiwanese Corporations in Globalisation and Regionalisation, by Gee San, April 1992.

Working Paper No. 62, Lessons from the Family Planning Experience for Community-Based Environmental Education, by Winifred Weekes-Vagliani, April 1992.

Working Paper No. 63, Mexican Agriculture in the Free Trade Agreement: Transition Problems in Economic Reform, by Santiago Levy and Sweder van Wijnbergen, May 1992.

Working Paper No. 64, Offensive and Defensive Responses by European Multinationals to a World of Trade Blocs, by John M. Stopford, May 1992.

Working Paper No. 65, Economic Integration in the Pacific Region, by Richard Drobnick, May 1992.

Working Paper No. 66, Latin America in a Changing Global Environment, by Winston Fritsch, May 1992.

Working Paper No. 67, An Assessment of the Brady Plan Agreements, by Jean-Claude Berthélemy and Robert Lensink, May 1992.

Working Paper No. 68, The Impact of Economic Reform on the Performance of the Seed Sector in Eastern and Southern Africa, by Elizabeth Cromwell, June 1992.

Working Paper No. 69, Impact of Structural Adjustment and Adoption of Technology on Competitiveness of Major Cocoa Producing Countries, by Emily M. Bloomfield and R. Antony Lass, June 1992.

Working Paper No. 70, Structural Adjustment and Moroccan Agriculture: an Assessment of the Reforms in the Sugar and Cereal Sectors, by Jonathan Kydd and Sophie Thoyer, June 1992.

Document de travail No. 71, L'Allégement de la dette au Club de Paris : les évolutions récentes en perspective, par Ann Vourc'h, juin 1992.

Working Paper No. 72, Biotechnology and the Changing Public/Private Sector Balance: Developments in Rice and Cocoa, by Carliene Brenner, July 1992.

Working Paper No. 73, Namibian Agriculture: Policies and Prospects, by Walter Elkan, Peter Amutenya, Jochbeth Andima, Robin Sherbourne and Eline van der Linden, July 1992.

Working Paper No. 74, Agriculture and the Policy Environment: Zambia and Zimbabwe, by Doris J. Jansen and Andrew Rukovo, July 1992.

Working Paper No. 75, Agricultural Productivity and Economic Policies: Concepts and Measurements, by Yair Mundlak, August 1992.

Working Paper No. 76, Structural Adjustment and the Institutional Dimensions of Agricultural Research and Development in Brazil: Soybeans, Wheat and Sugar Cane, by John Wilkinson and Bernardo Sorj, August 1992.

Working Paper No. 77, The Impact of Laws and Regulations on Micro and Small Enterprises in Niger and Swaziland, by Isabelle Journard, Carl Liedholm and Donald Mead, September 1992.

Working Paper No. 78, Co-Financing Transactions between Multilateral Institutions and International Banks, by Michel Bouchet and Amit Ghose, October 1992.

Document de travail No. 79, Allégement de la dette et croissance: le cas mexicain, par Jean-Claude Berthélemy et Ann Vourc'h, octobre 1992.

Document de travail No. 80, Le Secteur informel en Tunisie: cadre réglementaire et pratique courante, par Abderrahman Ben Zakour et Farouk Kria, novembre 1992.

Working Paper No. 81, Small-Scale Industries and Institutional Framework in Thailand, by Naruemol Bunjongjit and Xavier Oudin, November 1992.

Working Paper No. 81a, Statistical Annex: Small-Scale Industries and Institutional Framework in Thailand, by Naruemol Bunjongjit and Xavier Oudin, November 1992.

Document de travail No. 82, L'Expérience de l'allégement de la dette du Niger, par Ann Vourc'h et Maina Boukar Moussa, novembre 1992.

Working Paper No. 83, Stabilization and Structural Adjustment in Indonesia: an Intertemporal General Equilibrium Analysis, by David Roland-Holst, November 1992.

Working Paper No. 84, Striving for International Competitiveness: Lessons from Electronics for Developing Countries, by Jan Maarten de Vet, March 1993.

Document de travail No. 85, Micro-entreprises et cadre institutionnel en Algérie, par Hocine Benissad, mars 1993.

Working Paper No. 86, Informal Sector and Regulations in Ecuador and Jamaica, by Emilio Klein and Victor E. Tokman, August 1993.

Working Paper No. 87, Alternative Explanations of the Trade-Output Correlation in the East Asian Economies, by Colin I. Bradford Jr. and Naomi Chakwin, August 1993.

Document de travail No. 88, La Faisabilité politique de l'ajustement dans les pays africains, par Christian Morrisson, Jean-Dominique Lafay et Sébastien Dessus, novembre 1993.

Working Paper No. 89, China as a Leading Pacific Economy, by Kiichiro Fukasaku and Mingyuan Wu, November 1993.

Working Paper No. 90, A Detailed Input-Output Table for Morocco, 1990, by Maurizio Bussolo and David Roland-Holst November 1993.

Working Paper No. 91, International Trade and the Transfer of Environmental Costs and Benefits, by Hiro Lee and David Roland-Holst, December 1993.

Working Paper No. 92, Economic Instruments in Environmental Policy: Lessons from the OECD Experience and their Relevance to Developing Economies, by Jean-Philippe Barde, January 1994.

Working Paper No. 93, What Can Developing Countries Learn from OECD Labour Market Programmes and Policies?, by Åsa Sohlman with David Turnham, January 1994.

Working Paper No. 94, Trade Liberalization and Employment Linkages in the Pacific Basin, by Hiro Lee and David Roland-Holst, February 1994.

Working Paper No. 95, Participatory Development and Gender: Articulating Concepts and Cases, by Winifred Weekes-Vagliani, February 1994.

Document de travail No. 96, Promouvoir la maîtrise locale et régionale du développement : une démarche participative à Madagascar, par Philippe de Rham et Bernard Lecomte, juin 1994.

Working Paper No. 97, The OECD Green Model: an Updated Overview, by Hiro Lee, Joaquim Oliveira-Martins and Dominique van der Mensbrugghe, August 1994.

Working Paper No. 98, Pension Funds, Capital Controls and Macroeconomic Stability, by Helmut Reisen and John Williamson, August 1994.

Working Paper No. 99, Trade and Pollution Linkages: Piecemeal Reform and Optimal Intervention, by John Beghin, David Roland-Holst and Dominique van der Mensbrugghe, October 1994.

Working Paper No. 100, International Initiatives in Biotechnology for Developing Country Agriculture: Promises and Problems, by Carliene Brenner and John Komen, October 1994.

Working Paper No. 101, Input-based Pollution Estimates for Environmental Assessment in Developing Countries, by Sébastien Dessus, David Roland-Holst and Dominique van der Mensbrugghe, October 1994.

Working Paper No. 102, Transitional Problems from Reform to Growth: Safety Nets and Financial Efficiency in the Adjusting Egyptian Economy, by Mahmoud Abdel-Fadil, December 1994.

Working Paper No. 103, Biotechnology and Sustainable Agriculture: Lessons from India, by Ghayur Alam, December 1994.

Working Paper No. 104, Crop Biotechnology and Sustainability: a Case Study of Colombia, by Luis R. Sanint, January 1995.

Working Paper No. 105, Biotechnology and Sustainable Agriculture: the Case of Mexico, by José Luis Solleiro Rebolledo, January 1995.

Working Paper No. 106, Empirical Specifications for a General Equilibrium Analysis of Labor Market Policies and Adjustments, by Andréa Maechler and David Roland-Holst, May 1995.

Document de travail No. 107, Les Migrants, partenaires de la coopération internationale : le cas des Maliens de France, par Christophe Daum, juillet 1995.

Document de travail No. 108, Ouverture et croissance industrielle en Chine: étude empirique sur un échantillon de villes, par Sylvie Démurger, septembre 1995.

Working Paper No. 109, Biotechnology and Sustainable Crop Production in Zimbabwe, by John J. Woodend, December 1995.

Document de travail No. 110, Politiques de l'environnement et libéralisation des échanges au Costa Rica : une vue d'ensemble, par Sébastien Dessus et Maurizio Bussolo, février 1996.

Working Paper No. 111, Grow Now/Clean Later, or the Pursuit of Sustainable Development?, by David O'Connor, March 1996.

Working Paper No. 112, Economic Transition and Trade-Policy Reform: Lessons from China, by Kiichiro Fukasaku and Henri-Bernard Solignac Lecomte, July 1996.

Working Paper No. 113, Chinese Outward Investment in Hong Kong: Trends, Prospects and Policy Implications, by Yun-Wing Sung, July 1996.

Working Paper No. 114, Vertical Intra-industry Trade between China and OECD Countries, by Lisbeth Hellvin, July 1996.

Document de travail No. 115, Le Rôle du capital public dans la croissance des pays en développement au cours des années 80, par Sébastien Dessus et Rémy Herrera, juillet 1996.

Working Paper No. 116, General Equilibrium Modelling of Trade and the Environment, by John Beghin, Sébastien Dessus, David Roland-Holst and Dominique van der Mensbrugghe, September 1996.

Working Paper No. 117, Labour Market Aspects of State Enterprise Reform in Viet Nam, by David O'Connor, September 1996.

Document de travail No. 118, Croissance et compétitivité de l'industrie manufacturière au Sénégal, par Thierry Latreille et Aristomène Varoudakis, octobre 1996.

Working Paper No. 119, Evidence on Trade and Wages in the Developing World, by Donald J. Robbins, December 1996.

Working Paper No. 120, Liberalising Foreign Investments by Pension Funds: Positive and Normative Aspects, by Helmut Reisen, January 1997.

Document de travail No. 121, Capital Humain, ouverture extérieure et croissance : estimation sur données de panel d'un modèle à coefficients variables, par Jean-Claude Berthélemy, Sébastien Dessus et Aristomène Varoudakis, janvier 1997.

Working Paper No. 122, Corruption: The Issues, by Andrew W. Goudie and David Stasavage, January 1997.

Working Paper No. 123, Outflows of Capital from China, by David Wall, March 1997.

Working Paper No. 124, Emerging Market Risk and Sovereign Credit Ratings, by Guillermo Larraín, Helmut Reisen and Julia von Maltzan, April 1997.

Working Paper No. 125, Urban Credit Co-operatives in China, by Eric Girardin and Xie Ping, August 1997.

Working Paper No. 126, Fiscal Alternatives of Moving from Unfunded to Funded Pensions, by Robert Holzmann, August 1997.

Working Paper No. 127, Trade Strategies for the Southern Mediterranean, by Peter A. Petri, December 1997.

Working Paper No. 128, The Case of Missing Foreign Investment in the Southern Mediterranean, by Peter A. Petri, December 1997.

Working Paper No. 129, Economic Reform in Egypt in a Changing Global Economy, by Joseph Licari, December 1997.

Working Paper No. 130, Do Funded Pensions Contribute to Higher Aggregate Savings? A Cross-Country Analysis, by Jeanine Bailliu and Helmut Reisen, December 1997.

Working Paper No. 131, Long-run Growth Trends and Convergence Across Indian States, by Rayaprolu Nagaraj, Aristomène Varoudakis and Marie-Ange Véganzonès, January 1998.

Working Paper No. 132, Sustainable and Excessive Current Account Deficits, by Helmut Reisen, February 1998.

Working Paper No. 133, Intellectual Property Rights and Technology Transfer in Developing Country Agriculture: Rhetoric and Reality, by Carliene Brenner, March 1998.

Working Paper No. 134, Exchange-rate Management and Manufactured Exports in Sub-Saharan Africa, by Khalid Sekkat and Aristomène Varoudakis, March 1998.

Working Paper No. 135, Trade Integration with Europe, Export Diversification and Economic Growth in Egypt, by Sébastien Dessus and Akiko Suwa-Eisenmann, June 1998.

Working Paper No. 136, Domestic Causes of Currency Crises: Policy Lessons for Crisis Avoidance, by Helmut Reisen, June 1998.

Working Paper No. 137, A Simulation Model of Global Pension Investment, by Landis MacKellar and Helmut Reisen, August 1998.

Working Paper No. 138, Determinants of Customs Fraud and Corruption: Evidence from Two African Countries, by David Stasavage and Cécile Daubrée, August 1998.

Working Paper No. 139, State Infrastructure and Productive Performance in Indian Manufacturing, by Arup Mitra, Aristomène Varoudakis and Marie-Ange Véganzonès, August 1998.

Working Paper No. 140, Rural Industrial Development in Viet Nam and China: A Study in Contrasts, by David O'Connor, September 1998.

Working Paper No. 141, Labour Market Aspects of State Enterprise Reform in China, by Fan Gang, Maria Rosa Lunati and David O'Connor, October 1998.

Working Paper No. 142, Fighting Extreme Poverty in Brazil: The Influence of Citizens' Action on Government Policies, by Fernanda Lopes de Carvalho, November 1998.

Working Paper No. 143, How Bad Governance Impedes Poverty Alleviation in Bangladesh, by Rehman Sobhan, November 1998.

Document de travail No. 144, La libéralisation de l'agriculture tunisienne et l'Union européenne: une vue prospective, par Mohamed Abdelbasset Chemingui et Sébastien Dessus, février 1999.

Working Paper No. 145, Economic Policy Reform and Growth Prospects in Emerging African Economies, by Patrick Guillaumont, Sylviane Guillaumont Jeanneney and Aristomène Varoudakis, March 1999.

Working Paper No. 146, Structural Policies for International Competitiveness in Manufacturing: The Case of Cameroon, by Ludvig Söderling, March 1999.

Working Paper No. 147, China's Unfinished Open-Economy Reforms: Liberalisation of Services, by Kiichiro Fukasaku, Yu Ma and Qiumei Yang, April 1999.

Working Paper No. 148, Boom and Bust and Sovereign Ratings, by Helmut Reisen and Julia von Maltzan, June 1999.

Working Paper No. 149, Economic Opening and the Demand for Skills in Developing Countries: A Review of Theory and Evidence, by David O'Connor and Maria Rosa Lunati, June 1999.

Working Paper No. 150, The Role of Capital Accumulation, Adjustment and Structural Change for Economic Take-off: Empirical Evidence from African Growth Episodes, by Jean-Claude Berthélemy and Ludvig Söderling, July 1999.

Working Paper No. 151, Gender, Human Capital and Growth: Evidence from Six Latin American Countries, by Donald J. Robbins, September 1999.

Working Paper No. 152, The Politics and Economics of Transition to an Open Market Economy in Viet Nam, by James Riedel and William S. Turley, September 1999.

Working Paper No. 153, The Economics and Politics of Transition to an Open Market Economy: China, by Wing Thye Woo, October 1999.

Working Paper No. 154, Infrastructure Development and Regulatory Reform in Sub-Saharan Africa: The Case of Air Transport, by Andrea E. Goldstein, October 1999.

Working Paper No. 155, The Economics and Politics of Transition to an Open Market Economy: India, by Ashok V. Desai, October 1999.

Working Paper No. 156, Climate Policy Without Tears: CGE-Based Ancillary Benefits Estimates for Chile, by Sébastien Dessus and David O'Connor, November 1999.

Document de travail No. 157, Dépenses d'éducation, qualité de l'éducation et pauvreté : l'exemple de cinq pays d'Afrique francophone, par Katharina Michaelowa, avril 2000.

Document de travail No. 158, Une estimation de la pauvreté en Afrique subsaharienne d'après les données anthropométriques, par Christian Morrisson, Hélène Guilmeau et Charles Linskens, mai 2000.

Working Paper No. 159, Converging European Transitions, by Jorge Braga de Macedo, July 2000.

Working Paper No. 160, Capital Flows and Growth in Developing Countries: Recent Empirical Evidence, by Marcelo Soto, July 2000.

Working Paper No. 161, Global Capital Flows and the Environment in the 21st Century, by David O'Connor, July 2000.

Working Paper No. 162, Financial Crises and International Architecture: A "Eurocentric" Perspective, by Jorge Braga de Macedo, August 2000.

Document de travail No. 163, Résoudre le problème de la dette : de l'initiative PPTE à Cologne, par Anne Joseph, août 2000.

Working Paper No. 164, E-Commerce for Development: Prospects and Policy Issues, by Andrea Goldstein and David O'Connor, September 2000.

Working Paper No. 165, Negative Alchemy? Corruption and Composition of Capital Flows, by Shang-Jin Wei, October 2000.

Working Paper No. 166, The HIPC Initiative: True and False Promises, by Daniel Cohen, October 2000.

Document de travail No. 167, Les facteurs explicatifs de la malnutrition en Afrique subsaharienne, par Christian Morrisson et Charles Linskens, octobre 2000.

Working Paper No. 168, Human Capital and Growth: A Synthesis Report, by Christopher A. Pissarides, November 2000.

Working Paper No. 169, Obstacles to Expanding Intra-African Trade, by Roberto Longo and Khalid Sekkat, March 2001.

Working Paper No. 170, Regional Integration In West Africa, by Ernest Aryeetey, March 2001.

Working Paper No. 171, Regional Integration Experience in the Eastern African Region, by Andrea Goldstein and Njuguna S. Ndung'u, March 2001.

Working Paper No. 172, Integration and Co-operation in Southern Africa, by Carolyn Jenkins, March 2001.

Working Paper No. 173, FDI in Sub-Saharan Africa, by Ludger Odenthal, March 2001

Document de travail No. 174, La réforme des télécommunications en Afrique subsaharienne, par Patrick Plane, mars 2001.

Working Paper No. 175, Fighting Corruption in Customs Administration: What Can We Learn from Recent Experiences?, by Irène Hors; April 2001.

Working Paper No. 176, Globalisation and Transformation: Illusions and Reality, by Grzegorz W. Kolodko, May 2001.

Working Paper No. 177, External Solvency, Dollarisation and Investment Grade: Towards a Virtuous Circle?, by Martin Grandes, June 2001.

Document de travail No. 178, Congo 1965-1999: Les espoirs déçus du « Brésil africain », par Joseph Maton avec Henri-Bernard Solignac Lecomte, septembre 2001.

Working Paper No. 179, Growth and Human Capital: Good Data, Good Results, by Daniel Cohen and Marcelo Soto, September 2001.

Working Paper No. 180, Corporate Governance and National Development, by Charles P. Oman, October 2001.

Working Paper No. 181, *How Globalisation Improves Governance*, by Federico Bonaglia, Jorge Braga de Macedo and Maurizio Bussolo, November 2001.

Working Paper No. 182, Clearing the Air in India: The Economics of Climate Policy with Ancillary Benefits, by Maurizio Bussolo and David O'Connor, November 2001.

Working Paper No. 183, Globalisation, Poverty and Inequality in sub-Saharan Africa: A Political Economy Appraisal, by Yvonne M. Tsikata, December 2001.

Working Paper No. 184, Distribution and Growth in Latin America in an Era of Structural Reform: The Impact of Globalisation, by Samuel A. Morley, December 2001.

Working Paper No. 185, Globalisation, Liberalisation, Poverty and Income Inequality in Southeast Asia, by K.S. Jomo, December 2001.

Working Paper No. 186, Globalisation, Growth and Income Inequality: The African Experience, by Steve Kayizzi-Mugerwa, December 2001. Working Paper No. 187, The Social Impact of Globalisation in Southeast Asia, by Mari Pangestu, December 2001.

Working Paper No. 188, Where Does Inequality Come From? Ideas and Implications for Latin America, by James A. Robinson, December 2001.

Working Paper No. 189, Policies and Institutions for E-Commerce Readiness: What Can Developing Countries Learn from OECD Experience?, by Paulo Bastos Tigre and David O'Connor, April 2002.

Document de travail No. 190, La réforme du secteur financier en Afrique, par Anne Joseph, juillet 2002.

Working Paper No. 191, Virtuous Circles? Human Capital Formation, Economic Development and the Multinational Enterprise, by Ethan B. Kapstein, August 2002.

Working Paper No. 192, Skill Upgrading in Developing Countries: Has Inward Foreign Direct Investment Played a Role?, by Matthew J. Slaughter, August 2002.

Working Paper No. 193, Government Policies for Inward Foreign Direct Investment in Developing Countries: Implications for Human Capital Formation and Income Inequality, by Dirk Willem te Velde, August 2002.

Working Paper No. 194, Foreign Direct Investment and Intellectual Capital Formation in Southeast Asia, by Bryan K. Ritchie, August 2002. Working Paper No. 195, FDI and Human Capital: A Research Agenda, by Magnus Blomström and Ari Kokko, August 2002.

Working Paper No. 196, Knowledge Diffusion from Multinational Enterprises: The Role of Domestic and Foreign Knowledge-Enhancing Activities, by Yasuyuki Todo and Koji Miyamoto, August 2002.

Working Paper No. 197, Why Are Some Countries So Poor? Another Look at the Evidence and a Message of Hope, by Daniel Cohen and Marcelo Soto, October 2002.

Working Paper No. 198, Choice of an Exchange-Rate Arrangement, Institutional Setting and Inflation: Empirical Evidence from Latin America, by Andreas Freytag, October 2002.

Working Paper No. 199, Will Basel II Affect International Capital Flows to Emerging Markets?, by Beatrice Weder and Michael Wedow, October 2002.

Working Paper No. 200, Convergence and Divergence of Sovereign Bond Spreads: Lessons from Latin America, by Martin Grandes, October 2002.

Working Paper No. 201, Prospects for Emerging-Market Flows amid Investor Concerns about Corporate Governance, by Helmut Reisen, November 2002.

Working Paper No. 202, Rediscovering Education in Growth Regressions, by Marcelo Soto, November 2002.

Working Paper No. 203, Incentive Bidding for Mobile Investment: Economic Consequences and Potential Responses, by Andrew Charlton, January 2003.

Working Paper No. 204, Health Insurance for the Poor? Determinants of participation Community-Based Health Insurance Schemes in Rural Senegal, by Johannes Jütting, January 2003.

Working Paper No. 205, China's Software Industry and its Implications for India, by Ted Tschang, February 2003.

Working Paper No. 206, Agricultural and Human Health Impacts of Climate Policy in China: A General Equilibrium Analysis with Special Reference to Guangdong, by David O'Connor, Fan Zhai, Kristin Aunan, Terje Berntsen and Haakon Vennemo, March 2003.

Working Paper No. 207, India's Information Technology Sector: What Contribution to Broader Economic Development?, by Nirvikar Singh, March 2003.

Working Paper No. 208, Public Procurement: Lessons from Kenya, Tanzania and Uganda, by Walter Odhiambo and Paul Kamau, March 2003.

Working Paper No. 209, Export Diversification in Low-Income Countries: An International Challenge after Doha, by Federico Bonaglia and Kiichiro Fukasaku, June 2003.

Working Paper No. 210, Institutions and Development: A Critical Review, by Johannes Jütting, July 2003.

Working Paper No. 211, Human Capital Formation and Foreign Direct Investment in Developing Countries, by Koji Miyamoto, July 2003.

Working Paper No. 212, Central Asia since 1991: The Experience of the New Independent States, by Richard Pomfret, July 2003.

Working Paper No. 213, A Multi-Region Social Accounting Matrix (1995) and Regional Environmental General Equilibrium Model for India (REGEMI), by Maurizio Bussolo, Mohamed Chemingui and David O'Connor, November 2003.

Working Paper No. 214, Ratings Since the Asian Crisis, by Helmut Reisen, November 2003.

Working Paper No. 215, Development Redux: Reflections for a New Paradigm, by Jorge Braga de Macedo, November 2003.

Working Paper No. 216, The Political Economy of Regulatory Reform: Telecoms in the Southern Mediterranean, by Andrea Goldstein, November 2003.

Working Paper No. 217, The Impact of Education on Fertility and Child Mortality: Do Fathers Really Matter Less than Mothers?, by Lucia Breierova and Esther Duflo, November 2003.

Working Paper No. 218, Float in Order to Fix? Lessons from Emerging Markets for EU Accession Countries, by Jorge Braga de Macedo and Helmut Reisen, November 2003.

Working Paper No. 219, Globalisation in Developing Countries: The Role of Transaction Costs in Explaining Economic Performance in India, by Maurizio Bussolo and John Whalley, November 2003.

Working Paper No. 220, Poverty Reduction Strategies in a Budget-Constrained Economy: The Case of Ghana, by Maurizio Bussolo and Jeffery I. Round, November 2003.

Working Paper No. 221, Public-Private Partnerships in Development: Three Applications in Timor Leste, by José Braz, November 2003.

Working Paper No. 222, Public Opinion Research, Global Education and Development Co-operation Reform: In Search of a Virtuous Circle, by Ida Mc Donnell, Henri-Bernard Solignac Lecomte and Liam Wegimont, November 2003.

Working Paper No. 223, Building Capacity to Trade: What Are the Priorities?, by Henry-Bernard Solignac Lecomte, November 2003.

Working Paper No. 224, Of Flying Geeks and O-Rings: Locating Software and IT Services in India's Economic Development, by David O'Connor, November 2003.

Document de travail No. 225, Cap Vert: Gouvernance et Développement, par Jaime Lourenço and Colm Foy, novembre 2003.

Working Paper No. 226, Globalisation and Poverty Changes in Colombia, by Maurizio Bussolo and Jann Lay, November 2003.

Working Paper No. 227, The Composite Indicator of Economic Activity in Mozambique (ICAE): Filling in the Knowledge Gaps to Enhance Public-Private Partnership (PPP), by Roberto J. Tibana, November 2003.

Working Paper No. 228, Economic-Reconstruction in Post-Conflict Transitions: Lessons for the Democratic Republic of Congo (DRC), by Graciana del Castillo, November 2003.

Working Paper No. 229, Providing Low-Cost Information Technology Access to Rural Communities In Developing Countries: What Works? What Pays? by Georg Caspary and David O'Connor, November 2003.

Working Paper No. 230, The Currency Premium and Local-Currency Denominated Debt Costs in South Africa, by Martin Grandes, Marcel Peter and Nicolas Pinaud, December 2003.

Working Paper No. 231, Macroeconomic Convergence in Southern Africa: The Rand Zone Experience, by Martin Grandes, December 2003.

Working Paper No. 232, Financing Global and Regional Public Goods through ODA: Analysis and Evidence from the OECD Creditor Reporting System, by Helmut Reisen, Marcelo Soto and Thomas Weithöner, January 2004.

Working Paper No. 233, Land, Violent Conflict and Development, by Nicolas Pons-Vignon and Henri-Bernard Solignac Lecomte, February 2004.

Working Paper No. 234, The Impact of Social Institutions on the Economic Role of Women in Developing Countries, by Christian Morrisson and Johannes Jütting, May 2004.

Document de travail No. 235, La condition desfemmes en Inde, Kenya, Soudan et Tunisie, par Christian Morrisson, août 2004.

Working Paper No. 236, Decentralisation and Poverty in Developing Countries: Exploring the Impact, by Johannes Jütting, Céline Kauffmann, Ida Mc Donnell, Holger Osterrieder, Nicolas Pinaud and Lucia Wegner, August 2004.

Working Paper No. 237, Natural Disasters and Adaptive Capacity, by Jeff Dayton-Johnson, August 2004.

Working Paper No. 238, Public Opinion Polling and the Millennium Development Goals, by Jude Fransman, Alphonse L. MacDonnald, Ida Mc Donnell and Nicolas Pons-Vignon, October 2004.

Working Paper No. 239, Overcoming Barriers to Competitiveness, by Orsetta Causa and Daniel Cohen, December 2004.

Working Paper No. 240, Extending Insurance? Funeral Associations in Ethiopia and Tanzania, by Stefan Dercon, Tessa Bold, Joachim De Weerdt and Alula Pankhurst, December 2004.

Working Paper No. 241, Macroeconomic Policies: New Issues of Interdependence, by Helmut Reisen, Martin Grandes and Nicolas Pinaud, January 2005.

Working Paper No. 242, Institutional Change and its Impact on the Poor and Excluded: The Indian Decentralisation Experience, by D. Narayana, January 2005.

Working Paper No. 243, Impact of Changes in Social Institutions on Income Inequality in China, by Hiroko Uchimura, May 2005.

Working Paper No. 244, Priorities in Global Assistance for Health, AIDS and Population (HAP), by Landis MacKellar, June 2005.

Working Paper No. 245, Trade and Structural Adjustment Policies in Selected Developing Countries, by Jens Andersson, Federico Bonaglia, Kiichiro Fukasaku and Caroline Lesser, July 2005.

Working Paper No. 246, Economic Growth and Poverty Reduction: Measurement and Policy Issues, by Stephan Klasen, (September 2005).

Working Paper No. 247, Measuring Gender (In)Equality: Introducing the Gender, Institutions and Development Data Base (GID),

by Johannes P. Jütting, Christian Morrisson, Jeff Dayton-Johnson and Denis Drechsler (March 2006).

Working Paper No. 248, Institutional Bottlenecks for Agricultural Development: A Stock-Taking Exercise Based on Evidence from Sub-Saharan Africa by Juan R. de Laiglesia, March 2006.

Working Paper No. 249, Migration Policy and its Interactions with Aid, Trade and Foreign Direct Investment Policies: A Background Paper, by Theodora Xenogiani, June 2006.

Working Paper No. 250, Effects of Migration on Sending Countries: What Do We Know? by Louka T. Katseli, Robert E.B. Lucas and Theodora Xenogiani, June 2006.

Document de travail No. 251, L'aide au développement et les autres flux nord-sud : complémentarité ou substitution ?, par Denis Cogneau et Sylvie Lambert, juin 2006.

Working Paper No. 252, Angel or Devil? China's Trade Impact on Latin American Emerging Markets, by Jorge Blázquez-Lidoy, Javier Rodríguez and Javier Santiso, June 2006.

Working Paper No. 253, Policy Coherence for Development: A Background Paper on Foreign Direct Investment, by Thierry Mayer, July 2006. Working Paper No. 254, The Coherence of Trade Flows and Trade Policies with Aid and Investment Flows, by Akiko Suwa-Eisenmann and Thierry Verdier, August 2006.

Document de travail No. 255, Structures familiales, transferts et épargne : examen, par Christian Morrisson, août 2006.

Working Paper No. 256, Ulysses, the Sirens and the Art of Navigation: Political and Technical Rationality in Latin America, by Javier Santiso and Laurence Whitehead, September 2006.

Working Paper No. 257, Developing Country Multinationals: South-South Investment Comes of Age, by Dilek Aykut and Andrea Goldstein, November 2006.

Working Paper No. 258, The Usual Suspects: A Primer on Investment Banks' Recommendations and Emerging Markets, by Sebastián Nieto-Parra and Javier Santiso, January 2007.

Working Paper No. 259, Banking on Democracy: The Political Economy of International Private Bank Lending in Emerging Markets, by Javier Rodríguez and Javier Santiso, March 2007.

Working Paper No. 260, New Strategies for Emerging Domestic Sovereign Bond Markets, by Hans Blommestein and Javier Santiso, April 2007

Working Paper No. 261, Privatisation in the MEDA region. Where do we stand?, by Céline Kauffmann and Lucia Wegner, July 2007.

Working Paper No. 262, Strengthening Productive Capacities in Emerging Economies through Internationalisation: Evidence from the Appliance Industry, by Federico Bonaglia and Andrea Goldstein, July 2007.

Working Paper No. 263, Banking on Development: Private Banks and Aid Donors in Developing Countries, by Javier Rodríguez and Javier Santiso, November 2007.

Working Paper No. 264, Fiscal Decentralisation, Chinese Style: Good for Health Outcomes?, by Hiroko Uchimura and Johannes Jütting, November 2007.

Working Paper No. 265, Private Sector Participation and Regulatory Reform in Water supply: the Southern Mediterranean Experience, by Edouard Pérard, January 2008.

Working Paper No. 266, Informal Employment Re-loaded, by Johannes Jütting, Jante Parlevliet and Theodora Xenogiani, January 2008.

Working Paper No. 267, Household Structures and Savings: Evidence from Household Surveys, by Juan R. de Laiglesia and Christian Morrisson, January 2008.

Working Paper No. 268, Prudent versus Imprudent Lending to Africa: From Debt Relief to Emerging Lenders, by Helmut Reisen and Sokhna Ndoye, February 2008.

Working Paper No. 269, Lending to the Poorest Countries: A New Counter-Cyclical Debt Instrument, by Daniel Cohen, Hélène Djoufelkit-Cottenet, Pierre Jacquet and Cécile Valadier, April 2008.

Working Paper No.270, The Macro Management of Commodity Booms: Africa and Latin America's Response to Asian Demand, by Rolando Avendaño, Helmut Reisen and Javier Santiso, August 2008.

Working Paper No. 271, Report on Informal Employment in Romania, by Jante Parlevliet and Theodora Xenogiani, July 2008.

Working Paper No. 272, Wall Street and Elections in Latin American Emerging Democracies, by Sebastián Nieto-Parra and Javier Santiso, October 2008.

Working Paper No. 273, Aid Volatility and Macro Risks in LICs, by Eduardo Borensztein, Julia Cage, Daniel Cohen and Cécile Valadier, November 2008.

Working Paper No. 274, Who Saw Sovereign Debt Crises Coming?, by Sebastián Nieto-Parra, November 2008.

Working Paper No. 275, Development Aid and Portfolio Funds: Trends, Volatility and Fragmentation, by Emmanuel Frot and Javier Santiso, December 2008.

Working Paper No. 276, Extracting the Maximum from EITI, by Dilan Ölcer, February 2009.

Working Paper No. 277, Taking Stock of the Credit Crunch: Implications for Development Finance and Global Governance, by Andrew Mold, Sebastian Paulo and Annalisa Prizzon, March 2009.

Working Paper No. 278, Are All Migrants Really Worse Off in Urban Labour Markets? New Empirical Evidence from China, by Jason Gagnon, Theodora Xenogiani and Chunbing Xing, June 2009.

Working Paper No. 279, Herding in Aid Allocation, by Emmanuel Frot and Javier Santiso, June 2009.

Working Paper No. 280, Coherence of Development Policies: Ecuador's Economic Ties with Spain and their Development Impact, by Iliana Olivié, July 2009.

Working Paper No. 281, Revisiting Political Budget Cycles in Latin America, by Sebastián Nieto-Parra and Javier Santiso, August 2009.

Working Paper No. 282, Are Workers' Remittances Relevant for Credit Rating Agencies?, by Rolando Avendaño, Norbert Gaillard and Sebastián Nieto-Parra, October 2009.

Working Paper No. 283, Are SWF Investments Politically Biased? A Comparison with Mutual Funds, by Rolando Avendaño and Ja vier Santiso, December 2009.

Working Paper No. 284, Crushed Aid: Fragmentation in Sectoral Aid, by Emmanuel Frot and Javier Santiso, January 2010.

Working Paper No. 285, The Emerging Middle Class in Developing Countries, by Homi Kharas, January 2010.